UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): February 4, 2008

Hudson Highland Group, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 0-50129 (Commission File Number) 59-3547281 (IRS Employer Identification No.)

560 Lexington Avenue, New York, New York 10022 (Address of principal executive offices, including zip code)

(212) 351-7300

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Dere-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry into a Definitive Material Agreement.

The information provided in Item 2.01 of this Current Report on Form 8-K is hereby incorporated by reference into this Item 1.01.

Item 2.01. Completion of Acquisition or Disposition of Assets.

On February 4, 2008, Hudson Highland Group, Inc. (the "Company") completed the sale (the "Sale") of substantially all of the assets of its engineering and technical staffing division (the "Business") to System One Holdings, LLC ("Buyer"). The Sale was pursuant to an Asset Purchase Agreement (the "Agreement"), dated February 4, 2008, by and among the Company, its subsidiaries Hudson Global Resources Holdings, Inc. and Hudson Global Resources Management, Inc. and Buyer. Buyer is controlled by Troy Gregory, the former Chairman of the Company's Legal practice group. Greg Lignelli, the former Executive Vice President of the Company's Energy, Scientific and Managed Staffing practice group, is an investor in Buyer.

At the closing of the Sale, the Company received from Buyer pursuant to the Agreement (i) \$10,988,000 in cash, subject to a post-closing net working capital adjustment, (ii) a subordinated note in the aggregate principal amount of \$5,000,000 with a five year maturity and (iii) a warrant to purchase 10% of the units of membership interests in Buyer. The Company has the right to receive an additional \$600,000 that has been deposited by Buyer into an escrow account upon resolution of certain liabilities. The Company also retained approximately \$3,600,000 of receivables of the Business. The aggregate amount of such consideration for the Business was determined as a result of a competitive auction process.

The description of the Agreement set forth above is qualified by reference to the Agreement filed as Exhibit 2.1 to this Current Report on Form 8-K and incorporated by reference herein. A copy of the press release the Company issued on February 4, 2008, announcing the Sale is filed as Exhibit 99.1 and incorporated by reference herein.

Item 9.01. Financial Statements and Exhibits.

- (a) Not applicable.
- (b) Pro forma financial information.

The pro forma financial information required to be filed pursuant to Item 9.01(b) of Form 8-K is attached hereto as Exhibit 99.2 and incorporated by reference herein.

- (c) Not applicable.
- (d) <u>Exhibits</u>. The following exhibits are being filed herewith:
 - 2.1 Asset Purchase Agreement, dated as of February 4, 2008, by and among Hudson Highland Group, Inc., Hudson Global Resources Holdings, Inc., Hudson Global Resources Management, Inc., and System One Holdings, LLC. [The schedules to this document are not being filed herewith. Hudson Highland Group, Inc. agrees to furnish supplementally a copy of any such schedule to the Securities and Exchange Commission upon request.]
 - 99.1 Press Release of Hudson Highland Group, Inc. issued on February 4, 2008.
 - 99.2 Unaudited pro forma financial information of Hudson Highland Group, Inc.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

HUDSON HIGHLAND GROUP, INC.

Date: February 4, 2008

By: /s/ MARY JANE RAYMOND

Mary Jane Raymond Executive Vice President and Chief Financial Officer

HUDSON HIGHLAND GROUP, INC.

Exhibit Index to Current Report on Form 8-K

Exhibit <u>Number</u> 2.1	Asset Purchase Agreement, dated as of February 4, 2008, by and among Hudson Highland Group, Inc., Hudson Global Resources Holdings, Inc., Hudson Global Resources Management, Inc., and System One Holdings, LLC. [The schedules to this document are not being filed herewith. Hudson Highland Group, Inc. agrees to furnish supplementally a copy of any such schedule to the Securities and Exchange Commission upon request.]
99.1	Press Release of Hudson Highland Group, Inc. issued on February 4, 2008.
00.0	

99.2 Unaudited pro forma financial information of Hudson Highland Group, Inc.

EXECUTION COPY

ASSET PURCHASE AGREEMENT

among

HUDSON HIGHLAND GROUP, INC.,

HUDSON GLOBAL RESOURCES HOLDINGS, INC., and

HUDSON GLOBAL RESOURCES MANAGEMENT, INC.

and

SYSTEM ONE HOLDINGS, LLC

Dated as of February 4, 2008

TABLE OF CONTENTS

		Page
1. PURC	HASE AND SALE OF ASSETS; ASSUMPTION OF LIABILITIES	1
1.1.	Transfer of Assets	1
1.2.	Excluded Assets	3
1.3.	Assumed Liabilities	4
1.4.	Excluded Liabilities	5
1.5.	Nonassignable Contracts and Rights	6
2. PAYM	ENT OF PURCHASE PRICE; NET WORKING CAPITAL ADJUSTMENT	7
2.1.	Purchase Price and Payment	7
2.2.	Determination of Net Working Capital; Post-Closing Adjustment	8
2.3.	Allocation of Purchase Price	10
3. REPRI	ESENTATIONS AND WARRANTIES	11
3.1.	Representations and Warranties of Parent	
3.2.	Representations and Warranties of Buyer	21
3.3.	Expiration of Representations and Warranties	23
3.4.	No Other Representations; Projections	23
4 COVE	NANTS PRIOR TO CLOSING	23
4.1.	[Intentionally Omitted]	23
4.2.	[Intentionally Omitted]	24
4.3.	Further Actions	24
	TIONAL COVENANTS	24
5.1. 5.2.	Post-Closing Access to Information and Assistance	24
5.2. 5.3.	Further Agreements	25 25
5.3. 5.4.	Assistance and Cooperation Regarding Taxes and Financial Statements Further Assurances	25
5.4. 5.5.	Bulk Sales Law.	20
5.5. 5.6.	Employee Matters	20
5.7.	Seller Noncompetition and Nonsolicitation	20
5.8.	Security Deposits	29
5.9.	Corporate Name	29
5.10.	Accounts Receivable	29
5.11.	Buyer Noncompetition and Nonsolicitation	30
5.12.	Sales Tax Matters	31
5.13.	Hershey Receivables	31
6. COND 6.1.	DITIONS PRECEDENT TO OBLIGATIONS OF BUYER	31 31
6.1. 6.2.	Accuracy of Representations; Performance of Obligations No Orders.	31
6.2. 6.3.	Consents.	32
0.5.	00136116.	32

i

6.4.	Closing Deliveries.	32
7. CONDI 7.1. 7.2. 7.3.	TIONS PRECEDENT TO OBLIGATIONS OF SELLERS Accuracy of Representations; Performance of Obligations No Orders. Closing Deliveries.	32 32 32 32
8. INDEM 8.1. 8.2. 8.3. 8.4. 8.5. 8.6.	INIFICATION Indemnification by Parent. Indemnification By Buyer Procedures Relating to Indemnification Between Buyer and Parent Procedures Relating to Indemnification for Third Party Claims Insurance and Tax Effect Exclusive Remedy	32 32 34 34 35 36 36
9. CLOSII 9.1. 9.2. 9.3.	NG Closing Date Documents to be Delivered by Sellers Documents to be Delivered by Buyer	37 37 37 38
10. [INTE	NTIONALLY OMITTED]	38
11.1. 11.2. 11.3. 11.4. 11.5. 11.6.	ELLANEOUS Publicity Fees and Expenses Notices Binding Effect; Benefit Assignability Amendment	38 39 39 40 40 40
11.11. 11.12. 11.13.	Waiver Exhibits and Schedules Knowledge Articles and Section Headings; Table of Contents Severability No Strict Construction Applicable Law	40 41 41 41 41 41 41 41
11.15. 11.16.	Jurisdiction; Venue; Waiver of Jury Trial Entire Agreement; Exclusivity of Agreement Counterparts Definitions and Other Interpretive Matters	41 42 42 42

ii

SCHEDULES

Schedule A	Restrictive Covenants Agreements
Schedule 1.1(a)	Transferred Real Property Leases
Schedule 1.1(b)	Tangible Personal Property
Schedule 1.1(d)	Information Systems and Software
Schedule 1.2(e)	Excluded Contracts
Schedule 1.2(n)	Excluded Assets
Schedule 1.3(d)	Current and Accrued Liabilities
Schedule 1.4(m)	Excluded Liabilities
Schedule 3.1(c)	Governmental Approvals or Notice; Conflicts with Instruments
Schedule 3.1(d)	Financial Statements
Schedule 3.1(e)	Leased Real Property
Schedule 3.1(f)	Liens and Encumbrances
Schedule 3.1(g)	Material Contracts
Schedule 3.1(h)	Legal Proceedings
Schedule 3.1(i)	Employees
Schedule 3.1(j)	Intellectual Property
Schedule 3.1(k)	Government Licenses, Permits and Related Approvals
Schedule 3.1(l)	Conduct of Business in Compliance with Regulatory Requirements
Schedule 3.1(m)	Employee Benefit Plans and Arrangements
Schedule 3.1(n)	Environmental Matters
Schedule 3.1(p)	Tax Matters
Schedule 3.1(q)	Absence of Changes or Events
Schedule 3.1(r)	Absence of Undisclosed Liabilities
Schedule 3.1(s)	Material Customers
Schedule 3.1(t)	Insurance
Schedule 3.2(c)	Governmental Approvals or Notice; Conflicts with Instruments
Schedule 3.2(f)	Buyer Capitalization
Schedule 5.6	Leave of Absence Employees
Schedule 5.7	Noncompetition
Schedule 6.3	Consents
Schedule 11.9	Knowledge of Parent

EXHIBITS

Exhibit A	Form of Management Restrictive Covenants Agreements
Exhibit B	Subordinated Loan Agreement
Exhibit C	Warrant
Exhibit D	Agreed Accounting Principles
Exhibit E	Financing Commitments
Exhibit F	Operating Agreement
Exhibit G	Bill of Sale, Assignment and Assumption Agreement
Exhibit H	Transition Services Agreement
Exhibit I	Trademark Assignment
Exhibit J	Intercreditor Agreement
Exhibit K	Escrow Agreement

iii

ASSET PURCHASE AGREEMENT

ASSET PURCHASE AGREEMENT, dated as of February 4, 2008 (this "<u>Agreement</u>"), among HUDSON HIGHLAND GROUP, INC., a Delaware corporation ("<u>Parent</u>"), HUDSON GLOBAL RESOURCES HOLDINGS, INC., a Delaware corporation and a wholly owned subsidiary of Parent ("<u>Holdings</u>"), HUDSON GLOBAL RESOURCES MANAGEMENT, INC., a Pennsylvania corporation and a wholly owned subsidiary of Holdings ("<u>Management</u>;" Parent, Holdings and Management are referred to herein collectively as "<u>Sellers</u>"), SYSTEM ONE HOLDINGS, LLC, a Delaware limited liability company ("<u>Buyer</u>"), and, solely for the purposes of <u>Section 5.11</u>, Troy Gregory.

RECITALS

WHEREAS, Sellers desire to sell to Buyer, and Buyer desires to purchase and assume from Sellers, substantially all of the assets and liabilities of the engineering and technical staffing division of Sellers (the "Business"), upon the terms and subject to the conditions of this Agreement;

WHEREAS, concurrently with the execution and delivery of this Agreement, the members of the management team of the Business identified on <u>Schedule A</u>, annexed hereto and made a part hereof, have entered into restrictive covenants agreements with Buyer, in the form of <u>Exhibit A</u>, which will become effective upon the Closing; and

WHEREAS, capitalized terms used but not defined in the context of the Section in which such terms first appear shall have the meaning set forth in Section 11.17(a).

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants, agreements and conditions hereinafter set forth, and intending to be legally bound hereby, the parties hereto agree as follows:

1. Purchase and Sale of Assets; Assumption of Liabilities

1.1. <u>Transfer of Assets</u>. On the basis of the representations, warranties, covenants and agreements and subject to the satisfaction or waiver of the conditions set forth in this Agreement, on the Closing Date and subject to the provisions of <u>Section 1.2</u>, Sellers shall sell, convey, assign, transfer and deliver to Buyer, and Buyer shall purchase and acquire from Sellers, all of Sellers' right, title and interest in the assets, rights, properties, claims, contracts, business and goodwill of Sellers at the Closing Date that are utilized primarily in the Business of every kind, nature, character and description, tangible and intangible, real, personal or mixed, wherever located, including, without limiting the foregoing, the following (hereinafter referred to collectively as the "<u>Assets</u>"):

(a) Rights under the leases with respect to real property leased by Sellers identified on <u>Schedule 1.1(a)</u>, together with all improvements and fixtures thereon;

(b) All office equipment, furniture, phone systems, personal computers, supplies and other tangible personal property (x) located and utilized in each office location identified on <u>Schedule 1.1(a)</u>, and/or (y) identified on <u>Schedule 1.1(b)</u>, in each case, which is owned by Sellers and utilized primarily in the Business;

(c) All Sellers Intellectual Property (other than Sellers Intellectual Property specifically identified as an Excluded Asset on Schedule 3.1(j));

(d) All management information systems and software related primarily to the ownership of the Assets, including, for the avoidance of doubt, all such systems and software set forth on <u>Schedule 1.1(d)</u>;

(e) Rights under all contracts, arrangements, agreements, purchase or sale commitments for materials and services, advertising and promotional agreements, leases and other agreements ("<u>Contracts</u>") to which a Seller is a party or by which a Seller is bound that are primarily related to the Business, whether or not entered into in the ordinary course of the Business, including those Contracts set forth on <u>Schedule 3.1(g)</u>, but excluding those Contracts listed on <u>Schedule 1.2(e)</u> (the Contracts described in <u>Section 1.1(a)</u> and this <u>Section 1.1(e)</u> are hereinafter referred to as the "<u>Assumed Contracts</u>");

(f) All licenses, permits or franchises issued by any federal, state or municipal authority relating primarily to the development, use, maintenance or occupation of the Business ("Business Permits");

(g) Accounts receivable of Sellers (whether or not billed) to the extent attributable to services of the Business, excluding all intercompany receivables and the Hershey Receivables;

(h) All prepaid expenses, deposits and other accruals and all rights to goods and services and all other economic benefits arising out of prepayments, payments in advance and deposits by Sellers, to the extent related primarily to the Business;

(i) Records to the extent related exclusively to, or containing information exclusively about, the Business or the other Assets described in this Section 1.1;

(j) All claims for collection, indemnity rights and other claims and causes of action arising out of occurrences before or after the Closing and other intangible rights and assets in each case relating primarily to the other Assets described in this <u>Section 1.1</u> or to the Assumed Liabilities;

(k) All rights and claims of the Sellers, whether mature, contingent or otherwise, against third parties, whether in tort, contract or otherwise, including, without limitation, causes of action, unliquidated rights and claims under or pursuant to all warranties, representations and guarantees made by manufacturers, suppliers or vendors, claims for refunds, rights of off-set and credits of all kinds and all other general intangibles, in each case, only to the extent relating primarily to or used primarily in connection with the Business; <u>provided</u>, <u>however</u>, that such rights and claims shall not include rights and claims established by this Agreement or the other agreements contemplated hereby; and

(l) All goodwill generated by or associated with the Business.

1.2. Excluded Assets. It is expressly understood and agreed that the Assets shall not include the following (collectively, the "Excluded Assets"):

(a) Any equity interest in Holdings or Management and any rights in each Seller's franchise to be a corporation, its charter, corporate seal, stock books, minute books and other corporate records relating to the corporate organization and capitalization of such Seller;

(b) Any of the Business' assets that are consumed, sold or disposed of in the ordinary course of business consistent with past practice;

(c) Any refunds or credits with respect to any Taxes paid or incurred by Sellers (plus any related interest received or due from the relevant taxing authority), any prepaid Taxes of Sellers and any other rights related to Taxes paid or incurred by the Sellers with respect to any Pre-Closing Tax Periods;

(d) Any assets of Sellers not utilized primarily in the Business and all rights of Sellers under this Agreement or related to the transactions contemplated by this Agreement;

(e) Sellers' right, title and interest in and to the Contracts listed on Schedule 1.2(e);

(f) All intercompany receivables and the Hershey Receivables of Sellers;

(g) Cash and cash equivalents or similar type investments, deposits in transit, certificates of deposit, treasury bills and other marketable securities of Sellers, whether or not reflected as assets of the Business;

(h) Any rights in or to the use of the name, mark, trade name, trademark or service mark incorporating "Hudson" and any corporate symbols or logos related thereto, except to the extent provided in <u>Section 5.8</u>;

(i) Any insurance policies, or rights under such policies, held by Sellers;

(j) Any rights of Sellers with respect to (i) any Benefit Plan or Benefit Arrangement that is a defined benefit or defined contribution retirement plan or (ii) any assets held under the trust agreement or other funding arrangement related to any Benefit Plan or Benefit Arrangement that is a defined benefit or defined contribution retirement plan;

(k) All Records of Sellers (i) relating to accounting or financial matters or to Taxes or associated returns other than Records transferred to Buyer pursuant to <u>Section 1.1(i)</u>, (ii) relating primarily to any of the other Excluded Assets described in this <u>Section 1.2</u> or (iii) that do not contain information exclusively about or relating exclusively to the Business or the Assets described in <u>Section 1.1</u>;

(l) All prepaid items (including security deposits), claims for collection, indemnity rights and other claims and causes of action arising out of occurrences before or after the Closing and other intangible rights relating primarily to the other Excluded Assets described in this <u>Section 1.2</u> or to the Excluded Liabilities described in <u>Section 1.4</u>, and all privileges relating thereto;

(m) All medical records of Business Employees that are not permitted by Law to be transferred to Buyer; and

(n) Any assets identified on <u>Schedule 1.2(n)</u>.

1.3. <u>Assumed Liabilities</u>. Subject to the terms and conditions of this Agreement, on the Closing Date, Buyer shall assume and agree to pay, perform and discharge when due, subject to the provisions of <u>Section 1.5</u>, all of the Liabilities of Sellers to the extent relating to the Business or the Assets, whether arising before or after the Closing Date, to the extent the same are unpaid, undelivered or unperformed on the Closing Date, but only (except in the case of <u>Section 1.3(c)</u>) to the extent such Liabilities are, in the case of Liabilities that GAAP requires to be included in financial statements, both (x) liabilities of the type included in or reflected on the Financial Statements or the Recent Balance Sheet, and (y) included in the Net Working Capital calculation (collectively, the "<u>Assumed Liabilities</u>"), including:

(a) All Liabilities arising under the Assumed Contracts;

(b) All Liabilities arising under the Business Permits;

(c) All Liabilities that arise on account of Buyer's conduct of the Business, use of the Assets and/or delivery of services by Buyer, in each case, on or after the Closing Date, including Liabilities associated with continuing the litigation set forth in Item 1 of <u>Schedule 3.1(h)</u>;

(d) All current Liabilities and accrued Liabilities (excluding Taxes described in <u>Section 1.4(a)</u>) identified on <u>Schedule 1.3(d)</u> and all other current Liabilities and accrued Liabilities (excluding Taxes described in <u>Section 1.4(a)</u>) of the type identified on <u>Schedule 1.3(d)</u> that are incurred in the ordinary course of business and included in the Net Working Capital calculation; and

(e) All other Liabilities included in the Net Working Capital calculation and not otherwise listed above.

Buyer is not assuming, nor shall it be deemed to have assumed, any other Liabilities of Sellers of any kind or nature whatsoever, except as expressly provided in this Agreement or any instrument delivered pursuant to <u>Section 9.3(b)</u>.

1.4. <u>Excluded Liabilities</u>. It is expressly understood and agreed that Assumed Liabilities shall not include the following Liabilities of Sellers (collectively, the "<u>Excluded Liabilities</u>"):

(a) All Liabilities for Taxes imposed on or relating to (i) any Seller (including, without limitation, any liability for Taxes of any person other than such Seller under Treas. Reg. Sec. 1.1502-6 or any comparable provision of state, local or foreign law, as a transferee or successor, by contract, or otherwise, including without limitation any indemnification obligation), the Business, or the Assets or the use thereof for any Pre-Closing Tax Period (or any portion thereof) other than liabilities included in the Net Working Capital calculation, and (ii) any Seller after the Closing Date;

(b) All intercompany Liabilities and Liabilities arising under or related to any indebtedness for borrowed money;

(c) Liabilities covered by the insurance policies of Sellers in effect on or prior to the Closing Date, but only to the extent Sellers receive proceeds thereunder from third party insurers; <u>provided</u>, <u>however</u>, that, after the Closing Date, if any third party insurer seeks recoupment, refund or return of such proceeds from Buyer, then any such Liability to such third party insurer shall be deemed an Excluded Liability;

(d) Liabilities arising under any special incentive or other bonus agreements or arrangements between any Seller and any of its employees relating exclusively to the consummation of the transactions contemplated by this Agreement, including without limitation liabilities under the Retention Bonus Agreements entered into and to-be-entered into between Parent and the persons identified on <u>Schedule 1.2(e)</u>, including without limitation liabilities for any severance payments thereunder;

(e) All Liabilities that do not arise primarily out of or relate primarily to the Business or the Assets except as otherwise provided herein;

(f) Any Liability arising under or otherwise relating to any Benefit Plan or Benefit Arrangement, none of which plans or arrangements are being assumed by Buyer, other than liabilities of the type listed on Exhibit B to <u>Schedule 1.3(d)</u> that are included in the Net Working Capital calculation;

(g) Liabilities arising under the Contracts listed on Schedule 1.2(e); and

(h) Any Liability for any claim incurred or asserted by a Business Employee prior to the Closing, regardless of whether such claim is made or filed before or after the Closing, including without limitation Liabilities based upon breach of employment contract, employment discrimination, wrongful termination, wage and hour or health and safety requirements, workers compensation, ERISA, the Consolidated Omnibus Budget Reconciliation Act, the Worker Adjustment Retraining Notification Act of 1988, as amended, the Occupational Safety and Health Act of 1970, as amended, constructive termination, wrongful termination, failure to give reasonable notice or pay-in-lieu-of-notice, severance pay, termination pay, medical (and related) claims of Business Employees covered under Sellers' health and welfare plans, workers' compensation

claims, and accrued vacation of Business Employees up to and including the Closing Date, other than liabilities of the type listed on Exhibit B to <u>Schedule</u> <u>1.3(d)</u> that are included in the Net Working Capital calculation;

(i) All Liabilities other than liabilities included in the Net Working Capital calculation relating to independent contractors or independent subcontractors of the Business issued a Form 1099 ("1099 Contractors") or independent contractors or employees hired or retained and paid by the Business on an hourly basis to perform services for the Business regardless of whether or not such independent contractors or employees are performing such services ("Bench Employees");

(j) All Liabilities relating to actions, suits, proceedings, disputes, claims or investigations arising primarily out of or related primarily to the Business or the Assets relating to actions taken, or conduct occurring, prior to the Closing Date, other than Liabilities associated with continuing the litigation set forth in Item 1 of <u>Schedule 3.1(h)</u>;

(k) All Liabilities of Sellers for any violation of or failure to comply with Laws or Orders;

(1) All Liabilities relating to any rebate amounts owing to any customers of the Business for any work performed by Sellers prior to the Closing Date to the extent such liabilities are not included in the Net Working Capital calculation;

(m) Any Liabilities identified on <u>Schedule 1.4(m)</u>; and

(n) All Liabilities of the Sellers, including all Liabilities relating to the operation or ownership of the Business or the Assets prior to the Closing Date, which are not otherwise expressly identified as an "Assumed Liability" in <u>Section 1.3</u>.

1.5. <u>Nonassignable Contracts and Rights</u>. Notwithstanding anything to the contrary in this Agreement, no Contracts, properties, rights or other assets of Sellers shall be deemed sold, transferred or assigned to Buyer pursuant to this Agreement if the attempted sale, transfer or assignment thereof to Buyer without the consent or approval of another party or Government Entity would be ineffective or would constitute a breach of Contract or a violation of any Law or would in any other way materially adversely affect the rights of Sellers (or Buyer as transferee or assignee) and such consent or approval is not obtained on or prior to the Closing Date. In such case, to the extent possible, (a) the beneficial interest in or to such Contracts, properties, rights or other assets (collectively, the "<u>Beneficial Rights</u>") shall in any event pass as of the Closing Date to Buyer under this Agreement; and (b) pending such consent or approval, and so long as Sellers transfer and turn over all Beneficial Rights with respect to each such Contract, Buyer shall assume or discharge the Liabilities of Sellers under such Beneficial Rights (to the extent such obligations are Assumed Liabilities) as agent for Sellers, and Sellers shall act as Buyer's agent in the receipt of any benefits, rights or interest received from the Beneficial Rights. Buyer and Sellers shall use their commercially reasonable efforts (provided that no party shall be required to expend any money (other than reasonable attorney's fees), incur any liability, commence any litigation or offer or grant any accommodation (financial or otherwise) to any third party), to obtain and secure any and all consents and approvals that may be necessary to

effect the legal and valid sale, transfer or assignment of the Contracts, properties, rights or other assets underlying the Beneficial Rights, including their formal assignment or novation, if advisable. Buyer and Sellers will make or complete such transfers as soon as reasonably possible and cooperate with each other in any other reasonable arrangement designed to provide for Buyer the Beneficial Rights including enforcement at the cost and for the account of Buyer of any and all rights of Sellers against the other party thereto arising out of the breach or cancellation thereof by such other party or otherwise, and to provide for the discharge of any Liability arising under such Contracts, properties, rights or other assets, to the extent such Liability constitutes an Assumed Liability. To the extent any Contract may not be transferred or assigned to Buyer by reason of the absence of any such consent, then, notwithstanding anything in this Agreement to the contrary, Buyer shall not be required to assume any Assumed Liabilities arising under such Contract other than as expressly set forth in clause (b) of the second sentence of this <u>Section 1.5</u>.

2. Payment of Purchase Price; Net Working Capital Adjustment

2.1. <u>Purchase Price and Payment</u>. In consideration for the Assets, and subject to the terms and conditions of this Agreement, Buyer shall on the Closing Date:

(a) assume the Assumed Liabilities as provided in <u>Section 1.3</u>;

(b) issue and deliver to Parent a Subordinated Secured Note in the aggregate principal amount of \$5,000,000 (the "Subordinated Note") pursuant to, and in the form set forth in, the Financing Agreement in the form attached hereto as <u>Exhibit B</u> (the "Subordinated Loan Agreement");

(c) issue and deliver to Parent a warrant in the form attached hereto as Exhibit C (the "Warrant");

(d) transfer cash to the Escrow Agent in the amount of \$600,000; and

(e) transfer cash to Parent in the amount of \$11,561,000 (the "Initial Cash Purchase Price"), subject to the following adjustments:

(i) An increase equal to the amount, if any, by which the Net Working Capital as reflected on the Estimated Closing Statement is greater than the Target Net Working Capital; or

(ii) A reduction equal to the amount, if any, by which the Net Working Capital as reflected on the Estimated Closing Statement is less than the Target Net Working Capital.

The amount of the Initial Cash Purchase Price payable pursuant to this <u>Section 2.1</u> shall be paid on the Closing Date by wire transfer of immediately available funds to an account Parent has designated, at least two (2) business days prior to the Closing Date, in writing to Buyer.

The value tendered by Buyer pursuant to this <u>Section 2.1</u>, as adjusted pursuant to the provisions of <u>Section 2.2</u>, shall be hereinafter referred to as the "<u>Purchase Price</u>".

2.2. Determination of Net Working Capital; Post-Closing Adjustment.

(a) <u>Estimated Closing Statement</u>. For purposes of determining an estimate of the Net Working Capital to be reflected on the Closing Statement and the Purchase Price payable by Buyer at the Closing, not less than three (3) business days prior to the Closing Date, Parent shall, in consultation with Buyer, prepare and deliver to Buyer a statement that shall represent Parent's reasonable estimate of the Closing Statement. In the event Buyer shall object to any of the information set forth on such statement or accompanying schedules as presented by Parent, the parties shall negotiate in good faith and attempt to agree on appropriate adjustments so that such statement and accompanying schedules reflect a reasonable estimate of the Closing Statement and of the Net Working Capital to be reflected on the Closing Statement, but in the absence of such agreement, the good faith determination of the statement by Parent shall control (the statement as agreed to by the parties pursuant to this <u>Section 2.2(a)</u>, or in the absence of such agreement, the statement as prepared and delivered by Parent, is herein referred to as the "<u>Estimated Closing Statement</u>"). In connection with the determination of the Estimated Closing Statement, Parent shall provide to Buyer such information and detail as Buyer shall reasonably request.

(b) <u>Closing Statement</u>. Within forty-five (45) days following the Closing, Parent shall prepare, or cause to be prepared, and deliver to Buyer an unaudited statement (the "<u>Closing Statement</u>"), which shall set forth the Net Working Capital as of the Closing Date, and, except as set forth on <u>Exhibit D</u>, shall be prepared (i) in a manner consistent with the preparation of the Recent Balance Sheet as to accounting methods, policies, practices and procedures, with consistent classifications, judgments and estimation methodologies, and otherwise in accordance with GAAP, as in effect on the date of such preparation, except that the Closing Statement shall reflect only the Current Assets and the Current Liabilities, and (ii) based only on the information relating to the content of the Closing Statement that is known to Buyer or Parent on the Closing Date or becomes known by Buyer or Parent prior to the date on which Parent delivers the Closing Statement to Buyer if such information relates to an event that occurred prior to the Closing Date.

(c) <u>Disputes Regarding Closing Statement</u>. Buyer shall, within thirty (30) days after the delivery by Parent of the Closing Statement, complete its review of the Net Working Capital derived from the Closing Statement. If Buyer determines that the Closing Statement has not been prepared in accordance with <u>Section 2.2(b)</u> and <u>Exhibit D</u>, then Buyer shall inform Parent on or before the last day of such thirty (30) day period by delivering a written notice to Parent ("<u>Buyer's Objection</u>") (i) setting forth a specific description of the basis of Buyer's Objection and the adjustments to Net Working Capital that Buyer believes should be made and (ii) only including objections based on mathematical errors or based on the Closing Statement not being prepared in accordance with <u>Section 2.2(b)</u> and <u>Exhibit D</u>. Parent shall then have thirty (30) days to review and respond to Buyer's Objection. Parent and Buyer shall seek in good faith to

resolve in writing any differences which they may have with respect to any matter specified in Buyer's Objection and Parent shall have full access to the working papers of Buyer prepared in connection with Buyer's preparation of Buyer's Objection. If Parent and Buyer are unable to resolve all of their disagreements with respect to the determination of the foregoing items within twenty (20) days following the completion of Parent's review of Buyer's Objection, then Parent and Buyer shall refer their remaining differences to Grant Thornton LLP or another nationally recognized firm of independent public accountants as to which Parent and Buyer mutually agree (the "CPA Firm"), who shall, acting as accounting experts and not as arbitrators, determine on the basis of the standards set forth in Section 2.2(b) and Exhibit D, and only with respect to the remaining accounting-related differences so submitted by Buyer to Parent (and not by independent review), whether and to what extent, if any, Net Working Capital as derived from the Closing Statement requires adjustment. In making such determination, the CPA Firm shall consider only those items or amounts in the Closing Statement and Parent's calculation of the Net Working Capital as to which Buyer has disagreed in Buyer's Objection duly delivered pursuant to this Section 2.2(c) and may not assign a value greater than the greatest positive or negative adjustment requested by a party and in no event shall the Net Working Capital be more than Parent's calculation of the Net Working Capital delivered pursuant to Section 2.2(b) or less than Buyer's calculation of the Net Working Capital delivered pursuant to this Section 2.2(c). In connection with the engagement of the CPA Firm, Parent and Buyer shall execute reasonable engagement letters with the CPA Firm. Parent and Buyer shall direct the CPA Firm to use its reasonable best efforts to render its determination within forty-five (45) days. The CPA Firm's determination shall be conclusive and binding upon Buyer and Parent. The fees and disbursements of the CPA Firm shall be shared equally by Buyer and Parent. Buyer and Parent shall make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants) relating to the Closing Statement and all other items reasonably requested by the CPA Firm. The "Adjusted Closing Statement" shall be (i) the Closing Statement in the event that (x) no Buyer's Objection is delivered to Parent during the thirty (30) day period specified above, or (y) Parent and Buyer so agree, (ii) the Closing Statement, adjusted in accordance with Buyer's Objection in the event that Parent does not respond to Buyer's Objection within the thirty (30) day period following receipt by Parent of Buyer's Objection, or (iii) the Closing Statement, as adjusted by either (x) the agreement of Parent and Buyer or (y) the CPA Firm.

(d) <u>Cooperation</u>. Each of Buyer and Parent agrees that, following the Closing, it will not take any actions with respect to its accounting books, records, policies and procedures of the Business that would obstruct or prevent the preparation of the Closing Statement. Buyer shall cooperate with Parent in the preparation of the Closing Statement including, but not limited to, (i) providing Parent and Parent's representatives with full access during normal business hours to the books, records (including work papers, schedules, memoranda and other documents), facilities and Business Employees, (ii) causing employees of the Business to provide Parent as promptly as practicable following the Closing Date (but in no event later than ten (10) days after the Closing Date) with normal year-end closing financial information for the Business for the period ending as of the close of business on the day immediately prior to the Closing Date, and

(iii) cooperating fully with Parent and Parent's representatives, including the provision on a timely basis of all other information necessary or useful in connection with the preparation of the Closing Statement. Buyer and its accountants shall have full access to all information used by Parent in preparing the Closing Statement, including the work papers of its accountants.

(e) <u>Adjustment Payment to Buyer</u>. In the event the Net Working Capital as derived from the Adjusted Closing Statement is less than the Net Working Capital as reflected on the Estimated Closing Statement, Parent shall make an adjustment payment to Buyer in an amount equal to the difference between (i) the Net Working Capital as reflected on the Estimated Closing Statement and (ii) the Net Working Capital as derived from the Adjusted Closing Statement. Any payment required by the first sentence of this <u>Section 2.2(e)</u> shall be made by Parent to Buyer, together with interest thereon at the annual rate published by *The Wall Street Journal* as the "prime rate" at large U.S. money center banks as of the Closing Date (the "<u>Applicable Rate</u>") calculated on the basis of the number of days elapsed from and including the Closing Date to and excluding the date of payment, in immediately available funds within five (5) business days after the determination of the Adjusted Closing Statement.

(f) <u>Adjustment Payment to Parent</u>. In the event the Net Working Capital as derived from the Adjusted Closing Statement is greater than the Net Working Capital as reflected on the Estimated Closing Statement, Buyer shall make an adjustment payment to Parent in an amount equal to the difference between (i) the Net Working Capital as reflected on the Estimated Closing Statement and (ii) the Net Working Capital as derived from the Adjusted Closing Statement. Any payment required by the first sentence of this <u>Section 2.2(f)</u> shall be made by Buyer to Parent, together with interest thereon at the Applicable Rate calculated on the basis of the number of days elapsed from and including the Closing Date to and excluding the date of payment, in immediately available funds within five (5) business days after the determination of the Adjusted Closing Statement.

2.3. <u>Allocation of Purchase Price</u>. The aggregate Purchase Price shall be allocated among the Assets for tax purposes on the basis of the relative fair market values of such properties as of the Closing Date. Such values, and the value of the aggregate Purchase Price, shall be reasonably determined by Buyer and delivered to Parent as soon as practicable after the Closing (and in any event within one hundred eighty (180) days following the Closing Date) and shall be subject to Parent's consent, which consent shall not be unreasonably withheld or conditioned. Sellers and Buyer will follow and use such allocation in all tax returns, filings or other related reports made by them to any Government Entities. To the extent that disclosures of this allocation are required to be made by the parties to the IRS under the provisions of Section 1060 of the Code, or any regulations thereunder, Buyer and Parent will disclose such reports to the other prior to filing with the IRS.

3. Representations and Warranties

3.1. <u>Representations and Warranties of Parent</u>. Each Seller, jointly and severally, represents and warrants to Buyer as follows:

(a) <u>Due Organization and Power</u>. Each of Parent and Holdings is duly organized and validly existing under the laws of Delaware and Management is duly organized and validly existing under the laws of Pennsylvania. Each Seller has the requisite power and authority to own, operate and lease its properties and to conduct the Business as now conducted. Each Seller has all requisite power and authority to enter into this Agreement and any other agreement contemplated hereby and to perform its obligations hereunder and thereunder, including the power and authority to convey good title to Buyer with respect to the Assets owned by each Seller. Each Seller is duly authorized, qualified or licensed to do business as a foreign corporation in each of the jurisdictions in which its right, title or interest in or to any of the Assets, or the conduct of the Business, requires such authorization, qualification or licensing, except where the failure to so qualify or to be in good standing would not, individually or in the aggregate, have a Material Adverse Effect.

(b) <u>Authorization and Validity of Agreement</u>. The execution, delivery and performance by each Seller of this Agreement and any other agreements contemplated hereby and the consummation by each Seller of the transactions contemplated hereby and thereby have been duly authorized by the board of directors of each Seller. No other corporate action is necessary for the authorization, execution, delivery and performance by each Seller of this Agreement and any other agreements contemplated hereby and the consummation by each Seller of the transactions contemplated hereby or thereby. This Agreement and any other agreements contemplated hereby have been, or will be prior to or at Closing, duly executed and delivered by each Seller, and each constitutes, or will when so executed and delivered constitute, a valid and legally binding obligation of each Seller, enforceable against it in accordance with its respective terms, except as enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar Laws relating to or affecting creditors rights generally or by general equitable principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(c) <u>No Governmental Approvals or Notices; No Conflict with Instruments</u>. Except as described in <u>Schedule 3.1(c)</u>, the execution, delivery and performance of this Agreement and any other agreements contemplated hereby by each Seller and the consummation by each Seller of the transactions contemplated hereby and thereby (i) will not violate (with or without the giving of notice or the lapse of time or both), or require any authorization, consent, approval, filing or notice under, any provision of any Law or Order applicable to such Seller, except for such violations the occurrence of which, and such consents, approvals, filings or notices the failure of which to obtain or make, would not, individually or in the aggregate, have a Material Adverse Effect and except for such consents, approvals, filings or notice requirements which become applicable solely as a result of the specific regulatory status of Buyer or any of its Affiliates, and (ii) will not conflict with, or result in the breach or termination of any provision of, or constitute a

default under, or result in the acceleration of the performance of the obligations of such Seller under, or result in the creation of the right to accelerate, terminate, modify or cancel, or result in the creation of a lien, charge or encumbrance upon a portion of the Assets pursuant to, or require any notice under, the charter or by-laws of such Seller or the express terms of any Contract to which such Seller is a party or by which such Seller or any of the Assets is bound, except for such conflicts, breaches, terminations, defaults, accelerations or liens which would not, individually or in the aggregate, have a Material Adverse Effect.

(d) <u>Financial Statements</u>. <u>Schedule 3.1(d)</u> contains (i) an unaudited balance sheet of the Business as of September 30, 2007 (the "<u>Recent Balance</u> <u>Sheet</u>") and an unaudited statement of income of the Business for the nine months then ended, and (ii) an unaudited balance sheet of the Business as of December 31, 2006 and statements of income for the Business for the fiscal years ended December 31, 2006 and 2005 (collectively, the "<u>Financial</u> <u>Statements</u>"). Except as set forth on <u>Schedule 3.1(d)</u>, the Financial Statements were prepared in accordance with GAAP, as in effect on the date of such Financial Statements and applied on a consistent basis in such financial statements (except as may be indicated in the notes or comments to such Financial Statements), and such Financial Statements and notes or comments fairly present, in all material respects, the financial position and results of operations of the Business as of their respective dates and for the respective periods covered thereby giving effect to certain estimated allocations and charges for services disclosed on <u>Schedule 3.1(d)</u>.

(e) <u>Leased Real Property</u>. <u>Schedule 3.1(e)</u> contains a list that is complete and correct in all material respects of the real property leased by Sellers that is used by the Business.

(f) Title to Properties; Absence of Encumbrances; Sufficiency of Assets.

(i) Except as set forth on <u>Schedule 3.1(f)</u>, all of the Assets owned by Sellers are held free and clear of all claims, liens, security interests, charges, mortgages, pledges, easements, leases, encumbrances, licenses or sublicenses, conditional sales or other title retention agreements (an "<u>Encumbrance</u>") other than Permitted Encumbrances. Upon payment for the Assets as contemplated herein, Sellers shall convey to Buyer all of Sellers' right, title and interest in and to the Assets free and clear of all Encumbrances, except in each case (i) as specifically set forth in <u>Schedule 3.1(f)</u>, (ii) for Encumbrances for current Taxes and assessments not yet due and payable or being contested in good faith by appropriate proceedings for which reserves have been included in the Financial Statements, and (iii) for Encumbrances that are not material to the Business (such exceptions, collectively, the "<u>Permitted Encumbrances</u>"). Upon consummation of the transactions contemplated hereby, except for Permitted Encumbrances, Buyer shall acquire good and marketable title to the Assets.

(ii) Except for services and assets to be provided through the Transition Services Agreement and except as set forth on <u>Schedule 3.1(f)</u>, the Assets are sufficient for the conduct of the Business immediately following the Closing in substantially the same manner as currently conducted.

(iii) Except with respect to the representations and warranties contained in this <u>Section 3.1(f)</u>, Buyer is acquiring the Assets AS IS, WHERE IS. SELLERS DISCLAIM ALL OTHER REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, WITH RESPECT TO THE DESIGN, CONDITION, CAPACITY, VALUE, UTILITY, PERFORMANCE OR QUALITY OF THE ASSETS, AND SELLERS MAKE NO IMPLIED WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE WITH RESPECT THERETO.

(g) <u>Material Contracts</u>. <u>Schedule 3.1(g)</u> sets forth a list as of the date of this Agreement of each of the following types of written Contracts to which Sellers are a party that relate primarily to the Business:

(i) Any employment Contract with any Business Employee that has future liability in excess of \$25,000 per annum and is not terminable by notice of not more than sixty (60) calendar days for a cost of less than \$25,000;

(ii) Any covenant not to compete that materially restricts the operation of the Business;

(iii) Any Contract to lease personal property which has future liability in excess of \$25,000 per annum and is not terminable by notice of not more than sixty (60) calendar days for a cost of less than \$25,000;

(iv) Any Contract for loaning any money or directly or indirectly guaranteeing Liabilities of others (other than endorsements for the purpose of collection, loans made to employees for relocation, travel or other employment-related purposes, purchases of equipment or materials made under conditional sales contracts, in each case in the ordinary course of the Business), in each case having an outstanding principal amount or aggregate future liability (excluding interest) in excess of \$25,000;

(v) Any Contract under which any other person has directly or indirectly guaranteed Liabilities of Sellers (other than endorsements for the purpose of collection in the ordinary course of the Business), in each case having an outstanding principal amount or aggregate future liability (excluding interest) in excess of \$25,000;

(vi) Any other Contract, in each case of a type not described in any of <u>clauses (i)</u> through (v) above (without reference to disclosure thresholds set forth therein), to which Sellers are a party or by or to which any of their respective assets are bound or subject which has future liability in excess of \$50,000 per annum and is not terminable by such Seller by notice of not more than sixty (60) calendar days for a cost of less than \$50,000 (other than Contracts of Sellers in the ordinary course of the Business consistent with past practice); or

(vii) Any Contract involving the obligation of Sellers to deliver services to a customer of Sellers for which the customer will have an obligation under the Contract for payment after the date of this Agreement of more than \$200,000.

Sellers have made available to Buyer a copy of each Contract listed on <u>Schedule 3.1(g)</u> as amended to date. Except as disclosed on <u>Schedule 3.1(g)</u>, to the knowledge of Parent, each Contract described on <u>Schedule 3.1(e)</u> and <u>Schedule 3.1(g)</u> (collectively, the "<u>Material Contracts</u>") is valid, binding and in full force and effect and is enforceable by the applicable Seller in accordance with its terms. Except as disclosed in <u>Schedule 3.1(g)</u> or the other Schedules hereto, Sellers have, and to the knowledge of Parent, each counterparty thereto has, performed all material obligations required to be performed by them to date under the Material Contracts and Sellers are not, and to the knowledge of Parent, each counterparty thereto is not (with or without the lapse of time or the giving of notice, or both), in material breach or default thereof. As of the date hereof, no Seller has received any written or, to the knowledge of Parent, oral notice that any counterparty to any Material Contract threatened to terminate, suspend or not renew any Material Contract.

(h) <u>Legal Proceedings</u>. Except as described in <u>Schedule 3.1(h)</u>, there is no litigation, proceeding or governmental investigation pending or, to the knowledge of Parent, threatened (in a reasonably serious manner in writing) to which any Seller is a party relating primarily to the Assets or the Business or the transactions contemplated by this Agreement.

(i) Employees.

(i) <u>Schedule 3.1(i)</u> contains a true and complete list of:

(A) all employee handbooks, policies and manuals relating to the Business Employees of each Seller, copies of which have been delivered (or made available) to Buyer;

(B) all employees of the Business (each a "<u>Business Employee</u>"), together with (i) the job title, current rate of base salary or hourly wage, full- or part-time status, most recent annual bonus or commission, accrued sick, personal and vacation dates, date of hire and current employment status (e.g., leave of absence

and cause therefore, together with date that leave commenced and is expected to end) for each administrative, sales or back office Business Employee and (ii) the job description, employment status and employment class of each billable energy, engineering and Hershey Business Employee. Except as set forth on <u>Schedule 3.1(i)</u>, the employment of each Business Employee is at will;

(C) each consultant or independent contractor who currently provides services to the Business, including each 1099 Contractor and Benched Employee (each, a "Business Consultant"). A copy of each agreement between the Seller and each Business Consultant has been delivered (or made available) to Buyer.

(ii) Except as set forth in <u>Schedule 3.1(i)</u>, each Seller is in compliance in all material respects with all applicable Laws relating to the employment of each Business Employee, including without limitation, all Laws relating to wages, hours, employment standards, discrimination, safety and health, worker's compensation, and the Worker Adjustment and Retraining Notification Act of 1988 (the "<u>WARN Act</u>"), and any similar state, local or layoff statute. No claim by any past or present employee of the Business that such employee was subject to a wrongful discharge or any employment discrimination arising out of or relating to such employee's race, sex, age, religion, national origin, ethnicity, handicap or any other protected characteristic under applicable Laws is outstanding.

(iii) None of the Sellers nor any affiliate thereof, is a party to any agreement or contract with any union, labor organization or employee group which affects the employment of any Business Employee, including but not limited to, any collective bargaining agreements or labor contracts.

(iv) There has not been any strike, slowdown, picketing, work stoppage or labor dispute or, to the knowledge of Parent, any attempt by a labor union to organize the Business Employees.

(v) The Business has not, directly or through agents and independent contractors, employed any unauthorized aliens, as defined in 8 U.S.C. Section 1324a(h)(3). The Business has complied, or caused any such agent or independent contractor, to comply in all material respects with the employment verification and record-keeping requirements of 8 U.S.C. Section 1324a and 8 C.F.R. Section 274a, as amended. The Business is in compliance in all material respects with the Immigration and Nationality Act and the Immigration Reform and Control Act.

(vi) Except as set forth on <u>Schedule 3.1(i)</u>, there are no pending, or to the knowledge of Parent, threatened claims or actions as of the date of this Agreement by any Business Employee under any worker's compensation policy or long-term disability policy.

(vii) No "mass layoff" (as defined in the WARN Act), "plant closing" (as defined in such Act) or similar event has occurred with respect to the Business.

(viii) Except as set forth on <u>Schedule 3.1(i)</u>, there are no pending grievances, arbitrations or other disputes relating to Business Employees that would, individually or in the aggregate, have a Material Adverse Effect.

(j) Intellectual Property.

(i) <u>Schedule 3.1(j)</u> lists all Registered Intellectual Property owned by Sellers that is primarily used or primarily held for use in the Business as of the date hereof (the "<u>Sellers Intellectual Property</u>"). <u>Schedule 3.1(j)</u> also lists any proceedings or actions before any Government Entity (including the United States Patent and Trademark Office) related to any Sellers Intellectual Property.

(ii) Except as set forth in <u>Schedule 3.1(j)</u>, no claims are pending or, to the knowledge of Parent, threatened, against Sellers by any person with respect to the ownership, validity, enforceability, effectiveness or use in the Business of any Intellectual Property.

(iii) Except as set forth in <u>Schedule 3.1(j)</u>, all Sellers Intellectual Property that is material to the Business is fully transferable, alienable or licensable to Buyer by the Sellers without restriction and without payment of any kind to any third party. Each item of Sellers Intellectual Property is free and clear of any Encumbrances. Each Seller owns, or has acquired the necessary licenses to use, all Sellers Intellectual Property that are necessary to the operation or conduct of the Business as currently conducted and have been conducted within the twelve months prior to the date of this Agreement. No third party has any rights to any Sellers Intellectual Property owned by Sellers (other than non-exclusive license rights).

(iv) Except as set forth in <u>Schedule 3.1(j</u>), no Seller has received any written notice from any third party claiming that the operation of the Business or any act, product, technology or service of any Seller infringes, misappropriates, or dilutes any intellectual property right of any person.

(v) Except as set forth in <u>Schedule 3.1(j)</u>, no Sellers Intellectual Property owned by any Seller with respect to the Business is subject to any proceeding or outstanding order that restricts and/or conditions in any manner the use, transfer or licensing thereof by any Seller of such Sellers Intellectual Property.

(k) <u>Government Licenses, Permits and Related Approvals</u>. Except as described on <u>Schedule 3.1(k)</u>, Sellers have all licenses, permits, consents, approvals, authorizations, qualifications and orders of Government Entities required for the conduct of the Business as presently conducted by them, except insofar as any failure to have such licenses, permits, consents, approvals, authorizations, qualifications and orders would not, individually or in the aggregate, have a Material Adverse Effect.

(1) <u>Conduct of Business in Compliance with Regulatory Requirements</u>. Except as described on <u>Schedule 3.1(1)</u>, to the knowledge of Parent, Sellers are conducting the Business so as to comply with all applicable Laws, except insofar as any failure to comply with such Laws would not, individually or in the aggregate, have a Material Adverse Effect.

(m) Employee Benefit Plans and Arrangements.

(i) <u>Schedule 3.1(m)</u> hereto identifies:

(A) Each "employee benefit plan", as such term is defined in Section 3(3) of ERISA, that is maintained or otherwise contributed to by Sellers for the benefit of Business Employees and that is covered by ERISA (collectively, the "<u>Benefit Plans</u>"), copies of which have been provided or made available to Buyer.

(B) Each material Benefit Arrangement, copies of which have been provided or made available to Buyer.

(ii) Subject to the exceptions set forth on <u>Schedule 3.1(m</u>), each Benefit Plan and Benefit Arrangement has been maintained and administered at all times substantially in compliance with its terms and all applicable Laws, including ERISA and the Code, applicable to such Benefit Plan or Benefit Arrangement, except where the failure to do so would not reasonably be expected to result in a material liability to Buyer.

(iii) No Benefit Plan or Benefit Arrangement has terms requiring assumption by Buyer.

(iv) None of the Sellers nor any ERISA Affiliate currently has any obligation or liability with respect to any "defined benefit plan" as such term is defined in Section 3(35) of ERISA or to any "multiemployer plan" (as defined in Section 3(37) of ERISA). For purposes of this Agreement, the term "ERISA Affiliate" means each business, person or entity which would be treated as a "single employer" with either of the Sellers under Sections 414(b), (c), (m) or (o) of the Code or under Section 4001(b) of ERISA.

(v) There is no pending, or to the knowledge of Parent, threatened in writing, material claim, suit or other proceeding (other than ordinary and usual claims for benefits by participants and beneficiaries, including routine claims pursuant to domestic relations orders) with respect to any Benefit Plan or Benefit Arrangement that would have a material impact on this transaction or result in material liability to Buyer or result in the imposition of a material lien or other material claim against any of the Assets. To the knowledge of Parent, no Benefit Plan or Benefit Arrangement is under audit or investigation by the IRS, Department of Labor or other Government Entity or agency, nor has any such audit or investigation been threatened in writing.

(vi) Except as set forth in <u>Schedule 3.1(m</u>), neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will, either alone or in combination with another event, result in the payment to any Business Employee, 1099 Contractor, Benched Employee or Business Consultant of any money or other property or accelerate or provide any other rights or benefits to any Business Employee, 1099 Contractor, Benched Employee, 1099 Contractor, Benched Employee or Business Consultant.

(vii) Each Benefit Plan and Benefit Arrangement intended to be qualified under Section 401(a) of the Code is so qualified and is the subject of a currently effective favorable determination letter issued by the IRS with respect to the qualification of such Benefit Plan or Benefit Arrangement under the Code. No event has occurred, and no condition exists, which could adversely affect the tax-qualified status of any such Benefit Plan or Benefit Arrangement.

(viii) Each Seller has complied in good faith with the requirements of Section 409A of the Code with respect to each applicable Benefit Plan and Benefit Arrangements that covers Business Employees.

(n) <u>Environmental Matters</u>. Except as set forth on <u>Schedule 3.1(n)</u>, to the knowledge of Parent, Sellers' conduct of the Business complies with all applicable Laws regarding protection of the environment (including those protecting the quality of the ambient air, soil, surface water or groundwater, or those regarding the handling, storage, treatment, disposal, release or discharge of hazardous substances, or those requiring registration of tanks or permits for emissions to air, soil, surface water or ground water) in effect as of or, to the extent applicable, at any time prior to the date of this Agreement, except insofar as any failure to comply with such Laws would not, individually or in the aggregate, have a Material Adverse Effect.

(o) <u>Certain Fees</u>. Neither Sellers nor any of their respective officers, directors or employees, have employed any broker or finder or incurred any other liability for any brokerage fees, commissions or finders' fees in connection with the transactions contemplated hereby, except for those brokers whose fees will be paid by Parent.

(p) Taxes. Except where a failure to file Tax Returns, a failure of any such Tax Return to be complete and accurate in any respect or the failure to pay any Tax, individually or in the aggregate, would not have a Material Adverse Effect and except as set forth on Schedule 3.1(p): (i) each Seller has filed all Tax Returns required to be filed by it; (ii) all such Tax Returns are complete and accurate in all material respects; and (iii) all Taxes owed by each Seller (whether or not shown on any Tax Return) have been duly and timely paid, except with respect to matters contested in good faith in appropriate proceedings and disclosed to Buyer in writing. Except as set forth on <u>Schedule 3.1(p)</u>, (iv) there are no outstanding agreements or waivers extending any statute of limitations in respect of Taxes; (v) there is no action, suit, investigation, audit, claim or assessment pending or proposed, or, to the knowledge of any Seller, threatened, with respect to Taxes of any Seller; (vi) to the knowledge of Parent, no claim has ever been made by any Government Entity in a jurisdiction where a Seller does not file Tax Returns that such Seller is or may be subject to Taxes by such jurisdiction; (vii) all deficiencies asserted or assessments made as a result of any examination of any Tax Return of any Seller have been paid in full; (viii) there are no Encumbrances upon any of any Seller's assets arising in connection with any failure (or alleged failure) to pay any Tax; (ix) each Seller (A) has complied in all material respects with all Laws relating to the payment and withholding of Taxes from wages, salaries, or other payments to any employee or independent contractor of the Business; (B) has paid over to the proper Government Entity all amounts required to be so withheld; and (C) is not liable for any Taxes for failure to comply with such Laws; (x) none of the Sellers is a party to any Tax allocation or sharing agreement; (xi) none of the Assumed Liabilities is an obligation to make any payment that would not be deductible by reason of Section 280G of the Code; (xii) each of the Sellers has disclosed on its federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of federal income Tax within the meaning of Section 6662 of the Code; and (xiii) none of the Sellers has participated in any "reportable transaction" within the meaning of Treasury Regulation Section 1.6011-4(b).

(q) <u>Absence of Changes or Events</u>. Except as set forth on <u>Schedule 3.1(q)</u> or the other Schedules hereto, since the date of the Recent Balance Sheet, there has been no change, event, occurrence or circumstance with respect to the Business that has had or would have a Material Adverse Effect.

(r) <u>Absence of Undisclosed Liabilities</u>. Except as disclosed on <u>Schedule 3.1(r)</u>, to the knowledge of Parent, Sellers do not have any Liabilities relating to the Business that would constitute Assumed Liabilities, other than (i) Liabilities reflected on the Recent Balance Sheet, (ii) Liabilities that are not required by GAAP to be reflected on the Recent Balance Sheet, (iii) Liabilities under or arising out of the performance of Contracts disclosed on <u>Schedule 3.1(e)</u> or <u>Schedule 3.1(g)</u>, under or arising out of the performance of Contracts that are not required to be disclosed on <u>Schedule 3.1(g)</u> (but not Liabilities for breaches thereof) or under or arising out of the performance of Contracts entered into after the date of this Agreement in accordance with the terms and conditions of this Agreement, (iv) Liabilities disclosed in this Agreement or any other Schedule to this Agreement, or that are of the type or kind required to be disclosed in a Schedule to this Agreement but are not disclosed solely because they fall below the minimum threshold amount, term or materiality of the disclosures required by

the terms of this Agreement to be set forth in such Schedules and (v) Liabilities that have arisen since the date of the Recent Balance Sheet in the ordinary course of business consistent with past practice or otherwise in accordance with the terms and conditions of this Agreement.

(s) <u>Material Customers</u>. The revenues generated by Sellers from each of the customers identified on <u>Schedule 3.1(s)</u> in the nine month period ended September 30, 2007 and each of the fiscal years ended December 31, 2006, 2005 and 2004, is set forth opposite such customer's name on <u>Schedule 3.1(s)</u>.

(t) <u>Insurance</u>. The insurance policies maintained, as of the date of this Agreement, with respect to the Business and the Assets are set forth on <u>Schedule 3.1(t)</u>. All such policies are in full force and effect and all premiums due and payable thereon have been paid (other than retroactive or retrospective premium adjustments that are not yet, but may be, required to be paid with respect to any period ending prior to the Closing Date).

(u) 401(k) Plan. With respect to the Hudson Highland Group, Inc. 401(k) Savings Plan ("Parent's 401(k) Plan"):

(i) Parent has provided Buyer with a correct and complete copy (where applicable) of (A) Parent's 401(k) Plan, (B) each trust agreement or funding arrangement with respect to Parent's 401(k) Plan, (C) the three most recently filed annual reports on IRS Form 5500, (D) the most recently received IRS determination letter for Parent's 401(k) Plan, (E) the most recent summary plan description, and any summaries of material modification, of Parent's 401(k) Plan, (F) for the last three years, all material correspondence with the IRS, United States Department of Labor ("<u>DOL</u>") and any other Government Entity regarding Parent's 401(k) Plan, and (G) any other documents in respect of Parent's 401(k) Plan reasonably requested by Buyer.

(ii) (A) no actions or proceedings (other than routine claims for benefits in the ordinary course) are pending or, to the knowledge of Parent threatened, (B) no facts or circumstances exist that could reasonably be expected to give rise to any such actions or proceedings, and (iii) no administrative investigation, audit or other administrative proceeding by the DOL, the IRS or other Government Entity, including any voluntary compliance submission through the IRS's Employee Plans Compliance Resolution System or the DOL's Voluntary Fiduciary Correction Program, is pending, in progress or threatened.

(iii) Neither Parent, Sellers or any other "party in interest" or "disqualified person" with respect to Parent's 401(k) Plan has engaged in a nonexempt "prohibited transaction" within the meaning of Section 406 of ERISA or Section 4975 of the Code. No fiduciary of Parent's 401(k) Plan has any liability for breach of fiduciary duty or any other failure to act or

comply with the requirements of ERISA, the Code or any other applicable Laws in connection with the administration or investment of the assets of Parent's 401(k) Plan.

(iv) All contributions (including all employer contributions and employee salary reduction contributions) or premium payments required to have been made under Parent's 401(k) Plan as of the Closing have been made.

3.2. Representations and Warranties of Buyer. Buyer represents and warrants to Sellers as follows:

(a) <u>Due Organization and Power</u>. Buyer is a limited liability company duly organized and validly existing under the laws of Delaware. Buyer has all requisite power and authority to enter into this Agreement and any other agreement contemplated hereby and to perform its obligations hereunder and thereunder. Buyer is duly authorized, qualified or licensed to do business as a foreign limited liability company, and is validly existing, in each of the jurisdictions in which its right, title or interest in or to any asset, or the conduct of its business, requires such authorization, qualification or licensing, except where the failure to so qualify or to be in good standing would not have a material adverse effect on the ability of Buyer to perform its obligations hereunder or under any other agreement contemplated hereby.

(b) <u>Authorization and Validity of Agreement</u>. The execution, delivery and performance by Buyer of this Agreement and any other agreements contemplated hereby, including the Subordinated Loan Agreement, the Senior Loan Agreement, the Warrant and the LLC Agreement, and the consummation by Buyer of the transactions contemplated hereby and thereby have been duly authorized by its manager. No other company or member action is necessary for the authorization, execution, delivery and performance by Buyer of this Agreement and any other agreement contemplated hereby, including the Subordinated Loan Agreement, the Senior Loan Agreement, the Warrant and the LLC Agreement, and the consummation by Buyer of the transactions contemplated hereby. This Agreement has been, and the other agreements contemplated hereby, including the Subordinated Loan Agreement, the Warrant and the LLC Agreement, will be at the Closing, duly executed and delivered by Buyer, and each constitutes, or will when so executed and delivered constitute, a valid and legally binding obligation of Buyer, enforceable against Buyer in accordance with their respective terms, except as enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally or by general equitable principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(c) <u>No Governmental Approvals or Notices</u>; <u>No Conflict with Instruments</u>. Except as described in <u>Schedule 3.2(c)</u>, the execution, delivery and performance of this Agreement and any other agreements contemplated hereby, including the Subordinated Loan Agreement, the Senior Loan Agreement, the Warrant and the LLC Agreement, by Buyer and the consummation by it of the transactions contemplated hereby and thereby

(i) will not violate (with or without the giving of notice or the lapse of time or both), or require any consent, approval, filing or notice under, any provision of any Law or Order applicable to Buyer, except for such violations the occurrence of which, and such consents, approvals, filings or notices the failure of which to obtain or make, would not have a material adverse effect on Buyer's ability to perform its obligations hereunder, and (ii) will not conflict with, or result in the breach or termination of any provision of, or constitute a default under, or result in the acceleration of the performance of the obligations of Buyer, under, the certificate of formation or LLC Ageement or any indenture, mortgage, deed of trust, lease, licensing agreement, contract, instrument or other agreement to which Buyer is a party or by which Buyer or any of its assets or properties is bound, except for such conflicts, breaches, terminations, defaults, accelerations or liens which would not have a material adverse effect on Buyer's ability to perform its obligations hereunder.

(d) <u>Certain Fees</u>. Neither Buyer nor any of its officers, directors or employees, on behalf of Buyer, has employed any broker or finder or incurred any other liability for any brokerage fees, commissions or finders' fees in connection with the transactions contemplated hereby, except for those brokers whose fees will be paid by Buyer.

(e) <u>Financing</u>. Buyer has obtained legally binding equity financing commitments from the persons set forth in <u>Exhibit E</u> and Buyer and Brown Brothers have executed the Senior Loan Agreement, copies of each of which are attached as <u>Exhibit E</u> (the "<u>Financing Commitments</u>"), to provide cash to Buyer in an amount sufficient to enable Buyer to consummate the transactions contemplated by this Agreement. Subject to the Financing Commitments, Buyer (i) at the Closing will have sufficient funds available to pay the Purchase Price and all other payments required to be paid by Buyer pursuant to <u>Article 2</u>, and any expenses incurred by Buyer in connection with the transactions contemplated by this Agreement, (ii) at the Closing will have, the resources and capabilities (financial or otherwise) to perform Buyer's obligations hereunder, and (iii) has not incurred any obligation, commitment, restriction or Liability of any kind, which would impair or adversely affect such resources and capabilities.

(f) <u>Capitalization</u>. <u>Schedule 3.2(f)</u> sets forth the capitalization of Buyer as of the Closing. <u>Schedule 3.2(f)</u> sets forth the issued and outstanding equity interests of Buyer and the record and beneficial owners thereof as of the Closing. Except as set forth and described on <u>Schedule 3.2(f)</u>, there are no, and will not be at, or in connection with, the Closing, outstanding options, warrants or other rights to subscribe for or purchase, or securities convertible into or exchangeable for, equity interests of Buyer or any "phantom equity", equity appreciation right or similar instrument or arrangement which consists of or includes a right to acquire equity interests of Buyer or which is designed to create a right to any payment from Buyer or a claim against the assets of Buyer that is measured, in whole or in party, by reference to the value of an equity interest of Buyer (collectively, "<u>Equity Equivalents</u>"). Except as set forth and described on <u>Schedule 3.2(f)</u> or in the LLC Agreement, Buyer has not entered into any Contracts requiring it to issue any equity interests and Buyer has not made any commitments to issue any equity interests. Buyer has provided Parent with true, correct and complete copies of all Contracts to which Buyer or the holders of equity interests of Buyer are a party with respect to the voting, redemption, sale, transfer or other disposition of the equity interests of Buyer.

(g) <u>Warrant; LLC Agreement</u>. When issued to Parent upon the exercise of, and in accordance with the terms of, the Warrant, the membership interest of Buyer issued to Parent upon the exercise thereof shall be validly issued, fully paid and non-assessable membership interests of Buyer. <u>Exhibit F</u> is a true, correct and complete copy of the limited liability company agreement of the Buyer that will be in effect as of the Closing and pursuant to which the Seller Equity Interest will be issued (the "<u>LLC Agreement</u>").

3.3. Expiration of Representations and Warranties. The respective representations and warranties of Sellers and Buyer contained herein or in any certificate or other document delivered prior to or on the Closing Date shall survive the Closing, as follows: (a) the representations and warranties as to the matters set forth in Sections 3.1(a), 3.1(b), 3.1(f)(i) and 3.1(f)(ii) and Sections 3.2(a), 3.2(b), 3.2(e), 3.2(f) and 3.2(g) shall survive indefinitely, (b) the representations and warranties as to the matters set forth in Sections 3.1(m) and 3.1(p) shall survive until 45 days after expiration of the applicable statute of limitations, (c) the representations and warranties as to the matters set forth in Section 3.1(m) shall survive until three (3) years after the Closing Date, (d) all other representations and warranties set forth in Section 3.2 shall survive until the date that is eighteen (18) months after the Closing Date (the "Expiration Date"), and (e) all other representations and warranties set forth in Section 3.2 shall survive until the Expiration Date. The representations and warranties set forth in clause (a) above may be referred to herein as "Fundamental Representations".

3.4. <u>No Other Representations</u>; <u>Projections</u>. Except for the representations and warranties contained in <u>Section 3.1</u>, Buyer acknowledges that neither Sellers nor any other person or entity, acting on behalf of Sellers, makes or has made any other express or implied representation or warranty to Buyer as to the accuracy or completeness of any information regarding the Business or any other matter. Buyer has received or may receive from or on behalf of Sellers certain projections, including projected income statements and other financial information and certain business plan information for the Business. Buyer acknowledges that there are uncertainties inherent in attempting to make such estimates, projections and other forecasts and plans, that Buyer is familiar with the fact that, in the context of making projections, such uncertainties can exist, and that Buyer, therefore, is taking full responsibility for making its own evaluation of the adequacy and accuracy of all estimates, projections and other forecasts and plans so furnished to it (including the reasonableness of the assumptions underlying such estimates, projections and forecasts), and that Buyer shall have no claim against Sellers or any other person or entity acting on behalf of Sellers with respect thereto. Accordingly, Sellers make no representation or warranty with respect to such estimates, projections and plans (including the reasonableness of the assumptions or the accuracy of the information underlying such estimates, projections and forecasts).

4. Covenants Prior to Closing

4.1. [Intentionally Omitted]

4.2. [Intentionally Omitted]

4.3. <u>Further Actions</u>. Subject to the terms and conditions hereof, Sellers and Buyer shall use their reasonable best efforts to take, or cause to be taken, all action and to do, or cause to be done, and to cooperate fully with each other with respect to, all things necessary, proper or advisable to consummate and make effective the transactions contemplated by this Agreement, including using all commercially reasonable efforts: (a) to obtain all licenses, permits, consents, approvals, authorizations, qualifications and orders of Government Entities and parties to Contracts with Sellers that are necessary for the consummation of the transactions contemplated hereby and (b) to effect all necessary registrations and filings; <u>provided</u>, <u>however</u>, that such assistance shall not include any requirement of Sellers to expend any money (other than reasonable attorney's fees), to commence any litigation or offer or grant any accommodation (financial or otherwise) to any other party.

5. Additional Covenants

5.1. Post-Closing Access to Information and Assistance.

(a) For a period of seven (7) years after the Closing Date, each party hereto shall provide, and shall cause its appropriate personnel to provide, when reasonably requested to do so by another party, access to all tax, financial and accounting Records and any other Records transferred to Buyer or retained by Sellers, as applicable, in accordance with this Agreement and the right to make copies or extracts therefrom at its expense, in each case, for the purpose of complying with Laws, financial reporting obligations, Tax or employment matters or any civil litigation matters. Neither party shall, nor shall it permit its Affiliates to, intentionally dispose of, alter or destroy any such Records without giving thirty (30) days' prior written notice to the other party and permitting the other parties hereto, at their expense, to examine, duplicate or repossess such Records.

(b) Each party agrees to cooperate with the other party in the preparation for and prosecution of the defense of any claim, action or cause of action arising out of or relating to any Liability relating to the Business that arose prior to the Closing and that, in the case of Buyer, has been assumed by Buyer, or, in the case of Sellers has been retained by Sellers, including by making available evidence within the cooperating party's control and persons needed as witnesses employed by the cooperating party, as reasonably needed for such defense. Except as provided in <u>Article 8</u>, the requesting party shall reimburse the cooperating party for its actual out-of-pocket costs relating to its cooperation under this <u>Section 5.1(b)</u> and for a pro rata portion of the salary (including fringe benefits, with such pro rata portion determined based upon the time spent in connection with such cooperation) and for travel and subsistence expenses directly relating to the cooperating party's employees who assist the requesting party.

(c) Notwithstanding the provisions of <u>Sections 5.1(a)</u> and <u>5.1(b)</u>, while the existence of an adversarial proceeding between the parties will not abrogate or suspend the provisions of this <u>Section 5.1</u>, as to Records or other information directly pertinent to such dispute, the parties may not utilize this <u>Section 5.1</u> but rather, absent agreement, must utilize the available rules of discovery.

5.2. <u>Further Agreements</u>. Sellers authorize and empower Buyer on and after the Closing Date to receive and open all mail received by Buyer relating to the Business or the Assets and to deal with the contents of such communications in any proper manner. Sellers shall promptly deliver to Buyer any mail or other communication received by them after the Closing Date pertaining to the Business or the Assets and any cash, checks or other instruments of payment to which Buyer is entitled. Buyer shall promptly deliver to Parent any mail or other communication received by it after the Closing Date pertaining to the Excluded Assets and Excluded Liabilities, and any cash, checks or other instruments of payment in respect thereof.

5.3. Assistance and Cooperation Regarding Taxes and Financial Statements.

(a) <u>Information</u>. Subject to <u>Section 5.1</u>, Buyer and Parent each agree to furnish or cause to be furnished to each other, promptly upon request, any information and assistance relating to the Assets and the Business as the requesting party reasonably requests in connection with the preparation of financial statements, the filing of any Tax Return, the preparation for any Tax audit by a Government Entity, the response to any inquiry by a Government Entity, the mailing or filing of any notice and the prosecution or defense of any claim, suit or proceeding relating to any Tax Return or any other filing required to be made with any Government Entity or any other matter related to Taxes. Subject to <u>Section 5.1</u>, the parties shall cooperate with each other in the conduct of any audit or other proceeding related to Taxes involving the Business or the Assets or other proceeding related to Taxes involving the Business or the Assets.

(b) <u>Liability for Taxes</u>. Except as otherwise set forth herein or agreed to by Buyer and Parent, whenever it is necessary to determine the liability for Taxes for a portion of a taxable period that begins before and ends after the Closing Date, the determination of the Taxes for the portion of the taxable period ending on, and the portion of the year or period beginning after, the Closing Date shall be determined by assuming that the taxable period ended at the close of business on the Closing Date. If the Closing Date shall occur before the Tax rate for the year of Closing is fixed by the appropriate taxing authority, the apportionment of any such Taxes shall be upon the basis of the Tax rate for the preceding year applied to the latest assessed valuation and shall be readjusted promptly after such Tax rates are known. The Tax proration amount due from Sellers at Closing shall be paid by Sellers to Buyer. Such obligation to readjust shall survive the Closing.

(c) <u>Sales and Transfer Taxes</u>. Parent and Buyer shall each bear 50% of any transfer, sales, purchase, or similar Tax under the Laws of any Government Entity arising out of or resulting from the purchase of the Assets.

(d) <u>Cooperation on Tax Matters</u>. Subject to <u>Section 5.1</u>, each party hereto shall cooperate fully, as and to the extent reasonably requested by the other party, in connection with the filing of Tax Returns pursuant to this <u>Section 5.3</u> and responding to any audit, litigation or other proceeding with respect to Taxes in respect of the Business.

Subject to Section 5.1, such cooperation shall include the retention and (upon the other party's request) the provision of records and information which are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Each Seller agrees to retain all books and records with respect to Tax matters pertinent to such Seller that were in the possession of such Seller at Closing relating to any Pre-Closing Tax Period until the expiration of the statute of limitations (and any extensions thereof) of that Pre-Closing Tax Period, and to abide by all record retention agreements entered into with any Government Entity. Each party further agrees, upon request, to use its commercially reasonable efforts to obtain any certificate or other document from any Government Entity or any other person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed (including, but not limited to, with respect to the transactions contemplated hereby).

5.4. <u>Further Assurances</u>. From time to time after the Closing Date, Sellers and Buyer will execute and deliver, or cause to be executed and delivered, such other instruments of conveyance, assignment, transfer and delivery and will take such other actions as the other may reasonably request in order to more effectively consummate the transactions contemplated hereby, such as to transfer, convey, assign, and deliver to Buyer any of the Assets, or enable Buyer to discharge the Assumed Liabilities, or enable Buyer to exercise and enjoy all rights, benefits and obligations of Sellers with respect thereto.

5.5. Bulk Sales Law. Buyer and Sellers agree to waive compliance by the other with the provisions of the bulk sales law of any jurisdiction.

5.6. Employee Matters.

(a) Each Seller shall terminate the employment of all Business Employees with the applicable Seller, effective upon the Closing Date, and simultaneous therewith Buyer shall offer employment to each Business Employee (or with respect to Business Consultants shall offer the opportunity to continue to provide consulting services to the business) effective upon the Closing Date on an "at will" basis at the same salary or hourly wage rate and upon substantially similar terms and conditions of employment or consultancy, as the case may be, as those applicable to each Business Employee or Business Consultant immediately prior to the Closing; provided, however, that Buyer may amend or modify such terms and conditions of employment or consultancy after the Closing as it shall determine in its sole discretion.

(b) Notwithstanding the foregoing, with respect to each Business Employee identified on <u>Schedule 5.6</u> who is on a leave of absence from the Business at the time of the Closing, such Business Employee shall remain an employee of the applicable Seller, unless and until such time as such Business Employee returns from such approved leave of absence, at which time such Seller shall terminate the employment of such Business Employee and Buyer shall offer employment to each Business Employee on an "at will" basis at the same salary or hourly wage rate and upon substantially similar terms and conditions of employment as those applicable to such Business Employee immediately prior to the Closing; <u>provided</u>, <u>however</u>, that Buyer may amend or modify such terms and conditions of employment after the Closing as it shall determine in its sole discretion.

(c) Effective on the Closing Date, each Seller shall, and hereby does, release all Business Employees from any employment, non-compete and/or confidentiality agreement previously entered into between such Seller and such Business Employees to the extent necessary to allow such Business Employees to serve Buyer.

(d) Parent and Buyer agree as follows with respect to the transitional matters concerning tax-qualified 401(k) plans for the benefit of the Business Employees:

(i) As soon as administratively practicable after the Closing Date, Buyer shall adopt and implement a tax-qualified 401(k) plan for the benefit of the Business Employees ("Buyer's 401(k) Plan").

(ii) As soon as administratively practicable after the adoption of Buyer's 401(k) Plan, Parent shall cause the account balances of each Business Employee who participates in Parent's 401(k) Plan, valued as of the date of transfer (the "<u>Transfer Date</u>"), to be transferred from the trust maintained under Parent's 401(k) Plan (the "<u>Parent 401(k) Trust</u>") to the trust maintained under Buyer's 401(k) Plan (the "<u>Buyer 401(k) Trust</u>"), in accordance with Section 414(l) of the Code. Any shares of Seller or Parent stock held in a Business Employee's account under Parent's 401(k) Plan shall be sold prior to the Transfer Date and the cash proceeds from such sale shall be transferred to Buyer 401(k) Trust and, unless the Buyer's 401(k) Plan provides otherwise, initially invested in the default investment fund maintained under Buyer's 401(k) Plan. All other transferred assets shall be transferred in kind based on the investment election (or default election) of a Business Employee under Parent's 401(k) Plan in effect immediately prior to the Transfer Date. Notwithstanding the foregoing, to the extent that any portion of such transferred accounts is the subject of an outstanding loan to a participant as of the Transfer Date, the trustee of the Parent 401(k) Trust shall assign the promissory notes evidencing such loans to the Buyer 401(k) Plan until such account is in fact transferred in accordance with the foregoing (the transfer of an account, the "<u>Transfer</u>"); such responsibility and liability to be solely that of Parent's 401(k) Plan (and Parent 401(k) Trust). Following the Transfer, Buyer's 401(k) Plan shall be solely responsible for the payment of the accounts that are Transferred. To the extent that any adjustments or refunds must be made with respect to accounts of Business Employees under Parent's 401(k) Plan in order for Parent's 401(k) Plan to satisfy any non-discrimination for the plan year ended December 31, 2007 (the "2007 Plan Year"), such adjustments or refunds shall be watent prior to the Transfer Date.

(iii) Parent shall promptly transfer to Buyer all records of the Parent's 401(k) Plan that Buyer may reasonably require to administer and verify all benefits Transferred to Buyer's 401(k) Plan. Without limiting the foregoing, Parent shall promptly transfer to Buyer any qualified domestic relations order as described in section 414(p) of the Code that pertains to any Transferred account.

(iv) The employer matching contributions accrued under Parent's 401(k) Plan for the 2007 Plan Year with respect to each Business Employee shall be satisfied through a contribution by Buyer to Buyer's 401(k) Plan within the time period required by the Code, but in no event later than April 15, 2008. Parent shall promptly furnish Buyer with a detailed statement indicating the amount of matching contribution to be allocated to the account of each Business Employee for the 2007 Plan Year.

(e) Up to and including the Closing Date, each Seller shall be responsible for providing any notice of layoff or plant closing, as may be required with respect to any Business Employees pursuant to the Federal Worker Adjustment and Retraining Notification Act of 1988, any successor federal law and any applicable state or local plant closing notification statute. Following the Closing Date, Buyer shall be responsible for providing any notice of layoff or plant closing, as may be required with respect to any Business Employees pursuant to the Federal Worker Adjustment and Retraining Notification Act of 1988, any successor federal law and any successor federal law and any applicable state or local plant closing notification statute.

5.7. Seller Noncompetition and Nonsolicitation.

(a) For a period of three years after the Closing, none of the Sellers shall, and shall cause each of its Affiliates not to, directly or indirectly:

(i) engage in any activities or businesses, or establish any new businesses, within the United States that offer or provide the services set forth on <u>Schedule 5.7</u> ("<u>Competing Activities</u>"), or otherwise assist any person or entity in any way to do, or attempt to do, any of the foregoing; or

(ii) solicit for employment or hire any employees of the Business, unless such employee's employment with the Business has been terminated by Buyer at least six months before any such solicitation or hiring; <u>provided</u>, <u>however</u>, that the foregoing shall not prohibit solicitations to the public in general by way of non-targeted advertising in newspapers of general circulation or hiring employees as a result of such solicitation.

(b) Notwithstanding the foregoing, <u>Section 5.7(a)</u> shall not prohibit Parent or any of its Affiliates from (i) owning up to 1% of the outstanding capital stock of any

entity if such stock is listed on a national securities exchange or is regularly traded in the over-the-counter market by a member of a national securities exchange; (ii) owning or operating the business which they currently conduct other than the Business; or (iii) acquiring (whether through acquisition of equity interests or assets, merger or other business combination) a business that derives less than 10% of its annual revenues from Competing Activities (a "<u>Competitive Business</u>"), provided, however, if the annual revenues of such portion of the Competitive Business at any time exceed \$15,000,000 (measured on a trailing twelve month basis), that portion of the Competitive Business engaged in Competing Activities shall be offered to Buyer, and Buyer shall have a period of 30 days to elect to purchase such portion of the Competitive Business, and upon such election, Parent and Buyer shall negotiate the sale thereof in good faith on such terms and conditions as they mutually agree, and if such sale is not consummated within 60 days after expiration of such 30-day period, then Parent shall divest or discontinue such portion of the Competing Business within twelve (12) months after such 60-day period.

(c) Notwithstanding any other provision of this Agreement, it is understood and agreed that the remedy of indemnity payments pursuant to <u>Article 8</u> and other remedies at law would be inadequate in the case of any breach of the covenants contained in this <u>Section 5.7</u>. Buyer shall be entitled to equitable relief, including the remedy of specific performance, with respect to any breach or attempted breach of such covenants.

5.8. <u>Security Deposits</u>. Within thirty (30) days following the Closing, Buyer will replace or reimburse Sellers for the security deposits in the aggregate amount of \$13,223.91 held by Sellers' landlords with respect to the leases identified on <u>Schedule 1.1(a)</u>.

5.9. <u>Corporate Name</u>. Buyer acknowledges that, except for the limited right granted by this <u>Section 5.9</u>, it has no rights in or to the use of the name, mark, trade name, trademark or service mark incorporating "Hudson" or any variations thereof and any corporate symbols or logos related thereto (collectively, the "<u>Hudson Marks</u>"). Parent hereby grants Buyer a nonexclusive, royalty-free right to use (without sublicense), only in connection with the conduct and operation of the Business for a period of up to six (6) months after the Closing Date, the Hudson Marks (a) in displays, signage and postings for the period after the Closing Date necessary to permit the removal of such names as promptly as is reasonably feasible, and only to the extent such displays, signage or postings exist on the Closing Date; (b) to the extent any such names, trade names, trademarks, service marks or logos appear on stationery, supplies or advertising and marketing materials on hand or on order as of the Closing Date, until such is exhausted; <u>provided</u>, <u>however</u>, that Buyer shall not represent or hold itself out as representing Parent or its Affiliates.

5.10. <u>Accounts Receivable</u>. From and after the Closing, Buyer shall have the right and authority to collect for its own account all accounts receivable and other related items that are included in the Assets and to endorse with the name of the applicable Seller on any checks or drafts received with respect to any such accounts receivable or such other related items. From and after the Closing, none of the Sellers nor any person or entities acting on their behalf shall have the right to collect for any of the Sellers' account such accounts receivable or other items without the prior written consent of Buyer. The Sellers shall promptly deliver to Buyer any cash or other property received directly or indirectly by any of them with respect to

such accounts receivable and such other related items. Notwithstanding the foregoing, Sellers shall deliver any such accounts receivable or other related items that are \$100,000 or greater within two (2) business days for the first sixty (60) days after the Closing and thereafter once a week. The Sellers shall, at the request of Buyer, execute and deliver to Buyer such additional instruments, documents and agreements as Buyer may reasonably request to assist Buyer in the collection of such accounts receivable and such other related items.

5.11. Buyer Noncompetition and Nonsolicitation.

(a) Buyer and Troy Gregory acknowledge and agree that (w) Buyer would be unable to consummate the transactions contemplated hereby without Parent providing financing to Buyer with the Subordinated Note, (x) Troy Gregory, as the majority equity owner of Buyer, will benefit substantially from Buyer consummating the transactions contemplated hereby, including Parent providing financing to Buyer with the Subordinated Note, (y) Parent would not enter into this Agreement and provide financing to Buyer with the Subordinated Note absent the provisions of this <u>Section 5.11</u>, and (z) this Agreement and Parent's provision of financing to Buyer with the Subordinated Note constitute good and sufficient consideration for the restrictive covenants of Buyer and Troy Gregory set forth in this <u>Section 5.11</u>. For a period until the later of (1) two years after the Closing or (2) the time that Buyer pays Parent the full aggregate principal amount of the Subordinated Note and accrued interest thereon, neither Buyer nor Troy Gregory shall, and shall cause each of their Affiliates not to, directly or indirectly:

(i) own, operate, manage, control, engage in, invest in or participate in, act as a consultant or advisor to or render services for any activities or businesses within the United States that offer or provide recruitment or staffing services to the legal industry, including as an employee, agent, advisor, consultant, stockholder, partner, investor, owner, director or partner, or otherwise assist any person or entity in any way to do, or attempt to do, any of the foregoing; or

(ii) hire any employees of Parent and its Affiliates that are employed in Parent's legal practice group on November 1, 2007, unless such employee's employment with Parent or its Affiliates has been terminated by Parent or its Affiliates for a period of six months before any such solicitation or hiring.

(b) Notwithstanding the foregoing, <u>Section 5.11(a)</u> shall not prohibit Buyer, Troy Gregory or any of their Affiliates from owning up to 1% of the outstanding capital stock of any entity if such stock is listed on a national securities exchange or is regularly traded in the over-the-counter market by a member of a national securities exchange.

(c) Notwithstanding any other provision of this Agreement, it is understood and agreed that the remedy of indemnity payments pursuant to <u>Article 8</u> and other remedies at law would be inadequate in the case of any breach of the covenants contained in this <u>Section 5.11</u>. Parent shall be entitled to equitable relief, including the remedy of specific performance, with respect to any breach or attempted breach of such covenants.

(d) If, at the time of enforcement of this <u>Section 5.11</u>, a court holds that any of the restrictions stated herein are unreasonable under circumstances then existing, the parties hereto agree that the maximum period, scope or geographical area deemed reasonable under such circumstances will be substituted for the stated period, scope or area as contained in this <u>Section 5.11</u>.

5.12. <u>Sales Tax Matters</u>. Parent shall use commercially reasonable efforts to resolve the matters set forth on Item 1 of <u>Schedule 3.1(p)</u> with the applicable Government Entities with due regard to preserving the Business and the relationship with the applicable customers.

5.13. <u>Hershey Receivables</u>. Following receipt of proceeds from Hershey Receivables, Buyer agrees to promptly forward or cause to be forwarded to Parent any and all proceeds from Hershey Receivables that are received by Buyer after the Closing Date. Notwithstanding the foregoing, Buyer shall deliver any such proceeds that are \$100,000 or greater within two (2) business days for the first sixty (60) days after the Closing and thereafter once a week. If, after the Closing Date, Buyer receives any payment from The Hershey Company or its affiliates and at the time of such payment there are outstanding Hershey Receivables, on the one hand, and there are outstanding accounts receivable to Buyer from The Hershey Company or its affiliates, on the other hand ("<u>Buyer Receivables</u>"), and the payment (a) does not indicate whether it is in respect of Hershey Receivables or Buyer Receivables or (b) indicates that it is in payment of both Hershey Receivables and Buyer Receivables without specifying the portion to be allocated to each, then Parent and Buyer shall consult with each other to determine the proper allocation of such payments; and, if they are unable to reach agreement on the proper allocation, such payment shall be applied so as to retire Hershey Receivables and Buyer Receivables in chronological order to the oldest accounts first based upon the period of time such accounts receivable have existed on the books of Sellers or Buyer, as the case may be. Without limiting the foregoing, Buyer shall, and shall cause its employees with a relationship with The Hershey Company or its affiliates to, cooperate with Parent and provide Parent reasonable assistance in Parent's efforts to collect the Hershey Receivables.

6. <u>Conditions Precedent to Obligations of Buyer</u>. The obligations of Buyer under this Agreement are subject to the satisfaction (or written waiver by Buyer) prior to or at the Closing of each of the following conditions:

6.1. Accuracy of Representations; Performance of Obligations. Except for those representations and warranties which are made as of a particular date, the representations and warranties of Sellers in this Agreement shall be true and correct (ignoring for this purpose any materiality or Material Adverse Effect qualifiers in such representations and warranties) on the date hereof, except for breaches of representations and warranties that, individually or in the aggregate, would not have a Material Adverse Effect. The representations and warranties of Sellers in such representations and warranties) as of such date, except for breaches of representations and warranties) as of such date, except for breaches of representations and warranties as of a particular date shall be true and correct (ignoring for this purpose any materiality or Material Adverse Effect qualifiers in such representations and warranties) as of such date, except for breaches of representations and warranties that, individually or in the aggregate, would not have a Material Adverse Effect. Sellers shall have performed in all material respects the agreements, covenants and obligations to be performed by them prior to the consummation of the Closing. At the Closing, an executive officer of the Parent shall deliver to the Buyer a compliance certificate which shall certify as to matters set forth in this <u>Section 6.1</u>.

6.2. <u>No Orders</u>. No final Order shall have been issued by any Government Entity that restrains, enjoins or otherwise prohibits the transactions contemplated hereby and remains in effect.

6.3. <u>Consents</u>. Sellers shall have obtained, and delivered to Buyer, in form and substance reasonably acceptable to Buyer, written consents from each of the persons or entities identified on <u>Schedule 6.3</u>.

6.4. <u>Closing Deliveries</u>. Buyer shall have received the documents and deliveries required to be delivered pursuant to <u>Sections 2.2(a)</u> and <u>9.2</u>.

7. <u>Conditions Precedent to Obligations of Sellers</u>. The obligations of Sellers under this Agreement are subject to the satisfaction (or waiver by Parent) prior to or at the Closing of each of the following conditions:

7.1. <u>Accuracy of Representations</u>; <u>Performance of Obligations</u>. Except for those representations and warranties which are made as of a particular date, the representations and warranties of Buyer in this Agreement shall be true and correct in all material respects (except with respect to those representations and warranties which are qualified as to materiality, which shall be true in all respects) on the date hereof. The representations and warranties of Buyer contained in this Agreement which are made as of a particular date shall be true and correct in all material respects (except with respect to those representations and warranties which are qualified as to materiality, which shall be true and correct in all material respects (except with respect to those representations and warranties which are qualified as to materiality, which shall be true in all respects) as of such date. Buyer shall have performed or complied with in all material respects the agreements, covenants and obligations to be performed or complied with by it prior to the consummation of the Closing. At the Closing, an executive officer of Buyer shall deliver to the Parent a compliance certificate which shall certify as to matters set forth in this <u>Section 7.1</u>.

7.2. <u>No Orders</u>. No final Order shall have been issued by any Government Entity that restrains, enjoins or otherwise prohibits the transactions contemplated hereby and remains in effect.

7.3. Closing Deliveries. Parent shall have received the documents and deliveries required to be delivered pursuant to Sections 2.1 and 9.3.

8. Indemnification

8.1. Indemnification by Parent.

(a) Subject to the terms and conditions of this <u>Article 8</u>, Parent shall indemnify and hold harmless Buyer and its owners, directors, officers, employees and controlling persons and Buyer's 401(k) Plan and the Buyer 401(k) Trust and their respective shareholders, directors, employees, trustees, fiduciaries, administrators and agents (the "<u>Buyer Indemnitees</u>") from and against all Losses asserted against, resulting to, imposed upon or incurred by any such person, directly or indirectly, by reason of or

resulting from (i) any breach of the representations and warranties made by Sellers in <u>Section 3.1</u> of this Agreement or in the certificate delivered by Parent pursuant to <u>Section 6.1</u>, (ii) any breach of any covenant by any of the Sellers contained in this Agreement, (iii) any Excluded Liabilities, (iv) any claim that relates to Parent's 401(k) Plan, the Parent 401(k) Trust or any Transferred accounts under the Parent 401(k) Trust, that is attributable to, or has as a basis, any act or omission or thing whatsoever, alleged to have occurred on or prior to the Transfer, or (v) the matters set forth in Item 8 of <u>Schedule 3.1(p)</u>.

(b) Parent's obligations under <u>Section 8.1(a)</u> shall be subject to the following limitations:

(i) Except with respect to breaches of Fundamental Representations, and except for breaches of representations as a result of fraud, gross negligence and willful misconduct, Parent shall not have any indemnification obligation for Losses under <u>Section 8.1(a)(i)</u> unless and until the aggregate of all Losses relating thereto for which Parent would, but for this <u>Section 8.1(b)(i)</u>, be required to indemnify the Buyer Indemnitees exceeds on a cumulative basis an amount equal to \$150,000, at which point Parent, subject to <u>Section 8.1(b)(ii)</u>, shall indemnify Buyer for such Losses, but only to the extent such Losses exceed \$100,000.

(ii) Except with respect to breaches of Fundamental Representations, and except for breaches of representations as a result of fraud, gross negligence and willful misconduct, Parent shall not have any indemnification obligation for Losses under <u>Section 8.1(a)(i)</u> to the extent the aggregate amount of Losses for which Parent would otherwise be liable exceeds \$4,000,000.

(iii) Parent shall not have any indemnification obligation to the extent any Losses under <u>Section 8.1(a)(i)</u> were included as a liability or reserve in the calculation of the Net Working Capital pursuant to <u>Section 2.2</u>.

(iv) Parent's indemnification obligations under <u>Section 8.1(a)(i)</u> shall terminate as to each representation and warranty upon the expiration thereof pursuant to <u>Section 3.3</u>; <u>provided</u>, <u>however</u>, that such obligations shall not terminate with respect to any item as to which a Buyer Indemnitee shall have, prior to the expiration of the applicable representation and warranty, previously made a claim by delivering a notice that constitutes an Indemnification Notice and complies with the requirements therefore, but only with respect to the content of, and on the basis set forth in, such Indemnification Notice.

8.2. Indemnification By Buyer.

(a) Subject to the terms and conditions of this <u>Article 8</u>, Buyer shall indemnify and hold harmless Sellers and their directors, officers, employees and controlling persons (the "<u>Parent Indemnitees</u>"), from and against all Losses asserted against, resulting to, imposed upon or incurred by any such person, directly or indirectly, by reason of or resulting from (i) any breach of the representations and warranties made by Buyer in <u>Section 3.2</u> of this Agreement or in the certificate delivered by Buyer pursuant to <u>Section 7.1</u>, (ii) any breach of any covenant by Buyer contained in this Agreement, (iii) any Assumed Liabilities, or (iv) Buyer's use of the Hudson Marks except to the extent permitted in this Agreement.

(b) Buyer's obligations under Section 8.2(a) shall be subject to the following limitations:

(i) Except with respect to breaches of Fundamental Representations, and except for breaches of representations as a result of fraud, gross negligence and willful misconduct, Buyer shall not have any indemnification obligation for Losses under <u>Section 8.2(a)(i)</u> unless and until the aggregate of all Losses relating thereto for which Sellers would, but for this <u>Section 8.2(b)(i)</u>, be required to indemnify the Parent Indemnitees exceeds on a cumulative basis an amount equal to \$150,000, at which point Buyer, subject to <u>Section 8.2(b)(ii)</u>, shall indemnify Sellers for such Losses, but only to the extent such Losses exceed \$100,000.

(ii) Except with respect to breaches of Fundamental Representations, and except for breaches of representations as a result of fraud, gross negligence and willful misconduct, Buyer shall not have any indemnification obligation for Losses under <u>Section 8.2(a)(i)</u> to the extent the aggregate amount of Losses for which Buyer would otherwise be liable exceeds \$4,000,000.

(iii) Buyer's indemnification obligations under <u>Section 8.2(a)(i)</u> shall terminate as to each representation and warranty upon the expiration thereof pursuant to <u>Section 3.3</u>; <u>provided</u>, <u>however</u>, that such obligations shall not terminate with respect to any item as to which a Parent Indemnitee shall have, prior to the expiration of the applicable representation and warranty, previously made a claim by delivering a notice that constitutes an Indemnification Notice and complies with the requirements therefore, but only with respect to the content of, and on the basis set forth in, such Indemnification Notice.

8.3. <u>Procedures Relating to Indemnification Between Buyer and Parent</u>. Following the discovery of any facts or conditions that could reasonably be expected to give rise to a Loss or Losses for which indemnification under this <u>Article 8</u> can be obtained, the party seeking indemnification under this <u>Article 8</u> (the "<u>Indemnified Party</u>") shall, as soon as possible (and in any event within thirty (30) days) thereafter, provide written notice to the party from

whom indemnification is sought (the "<u>Indemnifying Party</u>"), setting forth, to the extent then known, the specific facts and circumstances, in reasonable detail, relating to such Loss or Losses, the amount of Loss or Losses (or a non-binding, reasonable estimate thereof if the actual amount is not known or not capable of reasonable calculation) and the specific Section(s) of this Agreement upon which the party seeking indemnification is relying in seeking such indemnification ("<u>Indemnification Notice</u>"); <u>provided</u>, <u>however</u>, the failure to timely deliver an Indemnification Notice hereunder shall not reduce an Indemnifying Party's obligation to provide indemnification under this <u>Article 8</u>, except to the extent such Indemnifying Party is actually prejudiced by such delinquent delivery.

8.4. Procedures Relating to Indemnification for Third Party Claims.

(a) <u>Notice</u>. In order for an Indemnified Party to be entitled to any indemnification provided for under this Agreement arising out of or involving a claim or demand made by any other person or entity, including a Government Entity (a "<u>Third Party Claim</u>"), the Indemnified Party shall provide an Indemnification Notice to the Indemnifying Party relating to the Third Party Claim as soon as possible (and in any event within thirty (30) days) after receipt by such Indemnifying Party's obligation to provide indemnification under this <u>Article 8</u>, except to the extent such Indemnifying Party is actually prejudiced by such delinquent delivery. Thereafter, the Indemnified Party shall deliver to the Indemnifying Party relating to the Third Party shall deliver to the Indemnifying Party, within five (5) business days after the Indemnified Party's receipt thereof, copies of all notices and documents (including court papers) received by the Indemnified Party relating to the Third Party Claim; provide, however, the failure to timely deliver any such copies shall not reduce the Indemnifying Party's obligation to provide indemnifying Party is actually prejudiced by such delinquent delivery. Thereafter, the Indemnified Party shall deliver to the Indemnifying Party is obligation to provide indemnified Party shall deliver to the Indemnified Party relating to the Third Party Claim; provided, however, the failure to timely deliver any such copies shall not reduce the Indemnifying Party's obligation to provide indemnification under this <u>Article 8</u>, except to the extent such Indemnifying Party is actually prejudiced by such delinquent delivery.

(b) <u>Defense</u>. If a Third Party Claim is made against the Indemnified Party, the Indemnifying Party shall be entitled to participate in the defense thereof and, if the Indemnifying Party so chooses, to assume the defense thereof with counsel selected by the Indemnifying Party and reasonably satisfactory to the Indemnified Party. Should the Indemnifying Party so elect to assume and thereafter actively and in good faith defend the defense of a Third Party Claim, the Indemnifying Party shall not be liable to the Indemnified Party for legal expenses subsequently incurred by the Indemnified Party in connection with the defense thereof. If the Indemnifying Party assumes such defense, then the Indemnified Party shall have the right to participate in the defense thereof and to employ counsel, at its own expense, separate from the counsel employed by the Indemnifying Party, it being understood, however, that the Indemnifying Party shall control such defense. The Indemnifying Party shall be liable for the fees and expenses of counsel employed by the Indemnified Party for any period during which the Indemnifying Party has not assumed the defense thereof or actively and in good faith defend such Third Party Claim. If the Indemnifying Party chooses to defend any Third Party Claim, all the parties hereto shall cooperate in the defense or prosecution of such Third Party Claim. Such cooperation shall include the retention and (upon the Indemnifying Party's request) the provision to the Indemnifying Party of Records that are

reasonably relevant to such Third Party Claim, and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. If the Indemnifying Party, within a reasonable time after receipt of an Indemnification Notice relating to a Third Party Claim (but in no event more than five (5) business days), chooses not to assume defense of a Third Party Claim or fails to defend such Third Party Claim actively and in good faith, the Indemnified Party shall (upon further notice) have the right to undertake the defense, compromise or settlement of such Third Party Claim or consent to the entry of judgment with respect to such Third Party Claim on behalf of, and for the account and risk of, the Indemnifying Party, and the Indemnifying Party shall have no right to challenge the Indemnified Party's defense, compromise, settlement or consent to judgment.

8.5. <u>Insurance and Tax Effect</u>. The amount of any Losses for which indemnification is provided under this <u>Article 8</u> shall be net of any (a) any amounts actually recovered by the Indemnified Party pursuant to any indemnification by, or indemnification or other agreement with, any third party, (b) Tax benefits actually realized by the Indemnified Party with respect to such Losses, and (c) amounts actually recovered from insurance proceeds in respect of such Losses (in each case net of any costs incurred to recover such amounts). The party seeking indemnification under this <u>Article 8</u> shall use commercially reasonable efforts (but shall not have to commence any litigation or arbitration) to make a claim for recovery of the amounts described in <u>Section 8.5(b)</u> and <u>8.5(c)</u>. The parties acknowledge and agree that no right of subrogation shall accrue and inure to the benefit of any source of any such amounts hereunder.

8.6. Exclusive Remedy. Except for rights expressly provided in <u>Article 5.12</u> and <u>Article 7</u>, and except for claims or causes of action based upon fraud, gross negligence and willful misconduct, the indemnification provisions of this <u>Article 8</u> shall be the sole and exclusive remedy with respect to any and all claims arising out of or relating to Buyer's investigation of the Business, this Agreement, the negotiation and execution of this Agreement or any Contract entered into pursuant hereto (except to the extent otherwise expressly set forth therein) or the performance by the parties of its or their terms, and except as provided herein, no other remedy shall be had pursuant to any contract, misrepresentation, or other tort theory or otherwise by Buyer, Parent or Sellers and their respective officers, directors, employees, agents, Affiliates, attorneys, consultants, successors and assigns, all such remedies being hereby expressly waived to the fullest extent permitted under applicable Law. Without limitation, except for claims or causes of action based upon fraud, gross negligence and willful misconduct, the procedures set forth in this <u>Article 8</u> constitute the sole and exclusive remedy of the parties arising out of any breach or claimed breach of the representations and warranties set forth in <u>Article 3</u>. In addition to the foregoing, except for claims or causes of action based upon fraud, gross negligence and willful misconduct, the amount of indemnification obligations of Parent set forth in this <u>Article 8</u> shall be the maximum amount of indemnification obligations set forth hereunder and none of the parties shall be entitled to a rescission of this Agreement (or any related agreements) or any further indemnification rights or claims of any nature whatsoever, all of which each party hereby waives.

9. Closing

9.1. <u>Closing Date</u>. The consummation of the transactions provided for in this Agreement shall take place at the offices of Foley & Lardner LLP, 777 East Wisconsin Avenue, Milwaukee, Wisconsin, at 7:00 a.m., Eastern Time, on February 4, 2008, or at such other time, date and place as shall be agreed upon by Parent and Buyer. It is understood and agreed that the consummation of the transactions contemplated hereby will be deemed for purposes of this Agreement and for all other intents and purposes to have occurred at 12:01 a.m., Eastern Time, on February 4, 2008 (the "<u>Closing</u>") and the date on which the Closing occurs shall be the "<u>Closing Date</u>."

9.2. <u>Documents to be Delivered by Sellers</u>. At the Closing, Sellers shall deliver to Buyer the following documents, in each case duly executed or otherwise in proper form:

(a) <u>Compliance Certificate</u>. The certificate described in <u>Section 6.1</u>, duly executed by an executive officer of the Parent;

(b) <u>Certified Resolutions</u>. Certified copies of the resolutions of the Boards of Directors of Sellers authorizing and approving this Agreement and the consummation of the transactions contemplated by this Agreement;

(c) <u>Instruments of Conveyance and Transfer</u>. A bill of sale, assignment and assumption agreement, in the form attached hereto as <u>Exhibit G</u>, annexed hereto and made a part hereof ("<u>Bill of Sale</u>"), executed by each of the Sellers;

(d) <u>Transition Services Agreement</u>. A transition services agreement, in the form of <u>Exhibit H</u>, duly executed by Parent (the "<u>Transition Services</u> <u>Agreement</u>");

(e) <u>FIRPTA Certificates</u>. Affidavits of non-foreign status for each Seller that comply with Section 1445 of the Code and the regulations promulgated thereunder;

(f) <u>Trademark Assignment</u>. A trademark assignment, in the form of <u>Exhibit I</u>, duly executed by Hudson Global Resources America, Inc. (the "<u>Trademark Assignment</u>");

(g) Intercreditor Agreement. An intercreditor agreement, in the form of Exhibit J, duly executed by Parent (the "Intercreditor Agreement");

(h) Subordinated Loan Agreement. The Subordinated Loan Agreement, duly executed by Parent;

(i) Escrow Agreement. An escrow agreement, in the form of Exhibit K, duly executed by Parent (the "Escrow Agreement"); and

(j) <u>Other Documents</u>. All other documents, instruments or writings required to be delivered to Buyer prior to or at the Closing pursuant to this Agreement and such other certificates of authority and documents as Buyer may reasonably request.

9.3. <u>Documents to be Delivered by Buyer</u>. At the Closing, Buyer shall deliver to Sellers the following documents, in each case duly executed or otherwise in proper form:

(a) <u>Compliance Certificate</u>. The certificate described in <u>Section 7.1</u>, duly executed by an executive officer of the Buyer;

(b) <u>Subordinated Loan Agreement</u>. The Subordinated Loan Agreement, the Subordinated Note and the Collateral Documents (as defined in the Subordinated Loan Agreement), each duly executed by Buyer;

(c) <u>Warrant</u>. The Warrant, duly executed by Buyer.

(d) <u>Certified Resolutions</u>. A certified copy of the resolutions of the Board of Directors of Buyer authorizing and approving this Agreement and the consummation of the transactions contemplated by this Agreement;

(e) Instruments of Assumption. The Bill of Sale, duly executed by Buyer;

(f) Transition Services Agreement. The Transition Services Agreement, duly executed by Buyer;

(g) LLC Agreement. The LLC Agreement, duly executed by the parties thereto;

(h) <u>Trademark Assignment</u>. The Trademark Assignment, duly executed by Buyer;

(i) Intercreditor Agreement. The Intercreditor Agreement, duly executed by Buyer and Brown Brothers;

(j) Escrow Agreement. The Escrow Agreement, duly executed by Buyer and the Escrow Agent; and

(k) <u>Other Documents</u>. All other documents, instruments or writings required to be delivered to Parent prior to or at the Closing pursuant to this Agreement and such other certificates of authority and documents as Parent may reasonably request.

10. [Intentionally Omitted]

11. Miscellaneous

11.1. <u>Publicity</u>. Parent and Buyer agree that, from the date hereof through the Closing Date, no public release or announcement concerning the transactions contemplated hereby shall be issued or made by any party without the prior consent of the other party (which consent shall not be unreasonably withheld or conditioned), except (a) as such release or announcement may, in the judgment of the releasing party, be required by Law or any rule or regulation of any United States securities exchange on which the securities of the releasing party are listed, in which case the party required to make the release or announcement shall allow the

other party reasonable time to comment on such release or announcement in advance of such issuance, and (b) subject to Buyer's prior approval of the form and substance thereof, which approval shall not be unreasonably withheld, delayed or conditioned, that Sellers may make such announcements to their employees. Notwithstanding the foregoing, Buyer and Parent may each issue a press release on the Closing Date, provided that the party issuing the release shall allow the other party reasonable time to comment on such release in advance of such issuance. Sellers and Buyer agree to keep the terms of this Agreement confidential, except to the extent required by applicable Law or for financial reporting purposes and except that the parties may disclose such terms to their respective accountants and other representatives as necessary in connection with the ordinary conduct of their respective businesses (so long as such persons agree to keep the terms of this Agreement confidential).

11.2. <u>Fees and Expenses</u>. Except as otherwise stated in this Agreement or as set forth below and regardless of whether or not the transactions contemplated hereby are consummated, all expenses incurred by the parties hereto shall be borne solely and entirely by the party that has incurred such expenses.

11.3. <u>Notices</u>. Except as otherwise set forth in this Agreement, all notices, requests, demands and other communications which are required or may be given under this Agreement shall be in writing and shall be deemed to have been duly given if delivered personally, if sent by facsimile or mailed, first class mail, postage prepaid, return receipt requested, as follows:

(a) if to Sellers:

Hudson Highland Group, Inc. 10 South Wacker Drive, Suite 2600 Chicago, Illinois 60606 Attention: Latham Williams Fax: (312) 795-4299

with a copy to:

Foley & Lardner LLP 777 East Wisconsin Avenue Milwaukee, Wisconsin 53202-5367 Attention: Benjamin F. Garmer, III Fax: (414) 297-4900

(b) if to Buyer:

System One Holdings, LLC 503 Martindale, Suite 200 Pittsburgh, Pennsylvania 15212 Attention: Daniel J. Moran, CPA Fax: 412-995-4603

with a copy to:

Lowenstein Sandler PC 1251 Avenue of the Americas, 18th Floor New York, New York 10020 Attention: Steven E. Siesser, Esq. Fax: 973.597.2507

or to such other address as either party shall have specified by notice in writing to the other party. All such notices, requests, demands and communications shall be deemed to have been received on the date of personal delivery or telecopy or on the third business day after the mailing thereof.

11.4. <u>Binding Effect; Benefit</u>. This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and assigns. Nothing in this Agreement, expressed or implied, is intended to confer on any person other than the parties hereto or their respective successors and assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement.

11.5. <u>Assignability</u>. This Agreement shall not be assignable (by operation of law or otherwise) by Sellers without the prior written consent of Buyer or by Buyer without the prior written consent of Parent, and any attempted assignment without such consent shall be void and without effect; <u>provided</u>, <u>however</u>, that Buyer may assign its rights under <u>Section 5.7</u> to an entity that acquires or is assigned (by operation of law or otherwise) substantially all of the assets of the Business and Parent may assign its rights under <u>Section 5.11</u> to an entity that acquires or is assigned (by operation of law or otherwise) substantially all of the assets of Parent's and its Affiliates' legal recruitment and staffing business, in each case upon written notice to the other party but without the consent of such other party.

11.6. <u>Amendment</u>. This Agreement shall not be amended, supplemented or otherwise modified except by a written instrument executed by Parent and Buyer.

11.7. <u>Waiver</u>. No waiver by any party of any of the provisions hereof shall be effective unless explicitly set forth in writing and executed by the party so waiving. Except as provided in the preceding sentence, no action taken pursuant to this Agreement, including any investigation by or on behalf of any party, shall be deemed to constitute a waiver by the party taking such action of compliance with any representations, warranties, covenants or agreements contained herein, and in any documents delivered or to be delivered pursuant to this Agreement and in connection with the Closing hereunder. The waiver by any party hereto of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach. No failure on the part of any party to exercise, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of such right, power or remedy by such party preclude any other or further exercise thereof or the exercise of any other right, power or remedy.

11.8. <u>Exhibits and Schedules</u>. The Exhibits and Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one Schedule shall be deemed to have been disclosed on each other Schedule to the extent it is reasonably clear on the face of such disclosure that such disclosed information is applicable thereto. Sellers may include in the Schedules items that are not material in order to avoid any misunderstanding, and such inclusion, or any references to dollar amounts, shall not be deemed to be an acknowledgement or representation that such items are material, to establish any standard of materiality or to define further the meaning of such terms for purposes of this Agreement. No disclosure on a Schedule relating to a possible breach or violation of any Contract, Law or Order shall be construed as an admission or indication that breach or violation exists or has actually occurred. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

11.9. <u>Knowledge</u>. The term "<u>knowledge</u>" when used in the phrases "<u>to the knowledge of Parent</u>" or "<u>Parent has no knowledge</u>" or words of similar import shall mean, and shall be limited to, the actual knowledge of the individuals listed on <u>Schedule 11.9</u> and shall include only their actual present knowledge obtained in their respective capacities with Sellers.

11.10. <u>Articles and Section Headings; Table of Contents</u>. The Articles and Section headings contained in this Agreement and the Table of Contents to this Agreement are for reference purposes only and shall not affect the meaning or interpretation of this Agreement. All references in this Agreement to any "Section" or "Article" are to the corresponding Section or Article of this Agreement unless otherwise specified.

11.11. <u>Severability</u>. If any provision of this Agreement shall be declared by any court of competent jurisdiction to be illegal, void or unenforceable, then all other provisions of this Agreement shall not be affected and shall remain in full force and effect.

11.12. <u>No Strict Construction</u>. Notwithstanding the fact that this Agreement has been drafted or prepared by one of the parties, each of the parties confirms that both it and its counsel have reviewed, negotiated and adopted this Agreement as the joint agreement and understanding of the parties, and the language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any person.

11.13. <u>Applicable Law</u>. The interpretation and construction of this Agreement, and all matters relating hereto, will be governed by the laws of the State of New York applicable to contracts made and to be performed entirely within the State of New York without giving effect to any conflict of law provisions thereof.

11.14. Jurisdiction; Venue; Waiver of Jury Trial. Each party irrevocably and unconditionally submits to the exclusive jurisdiction of any State or Federal court sitting in New York County, State of New York over any suit, action or proceeding arising out of or relating to this Agreement. Without limitation of other means of service, each party agrees that service of any process, summons, notice or document with respect to any action, suit or proceeding may be

served on it in accordance with the notice provisions set forth herein. Each party irrevocably and unconditionally waives any objection to the laying of venue of any such suit, action or proceeding brought in any such court and any claim that such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. Each party agrees that a final judgment in any such suit, action or proceeding brought in any such court shall be conclusive and binding upon such party and may be enforced in any other courts to whose jurisdiction such party is or may be subject, by suit upon such judgment. Each party hereby waives its rights to a trial by jury of any claim or cause of action arising out of or relating to Buyer's investigation of the Business, this Agreement, the negotiation and execution of this Agreement or any Contract entered into pursuant hereto (except to the extent otherwise expressly set forth therein) or the performance by the parties of their obligations hereunder or thereunder in any action, proceeding or other litigation of any type brought by one party against the other, regardless of the basis of the claim or cause of action.

11.15. Entire Agreement; Exclusivity of Agreement. This Agreement (including the Exhibits and Schedules hereto) and the Confidentiality Agreement constitute the entire agreement between the parties hereto and supersede all prior agreements and understandings, oral and written, between the parties hereto with respect to the subject matter hereof. The parties hereto have voluntarily agreed to define their rights, liabilities and obligations respecting the subject matter of this Agreement exclusively in contract pursuant to the express terms and provisions of this Agreement. Except for claims for fraud and willful misconduct, the sole and exclusive remedies for any breach of the terms and provisions of this Agreement (including any representations and warranties set forth herein) shall be those remedies available at law or in equity for breach of contract only (as such contract remedies may be further limited or excluded pursuant to the express terms of this Agreement).

11.16. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which together shall be deemed to be one and the same instrument.

11.17. Definitions and Other Interpretive Matters.

(a) For purposes of this Agreement, the following terms shall have the meanings specified in this Section 11.17(a):

"1099 Contractor" shall have the meaning set forth in Section 1.4(i).

"2007 Plan Year" shall have the meaning set forth in Section 5.6(d).

"Adjusted Closing Statement" shall have the meaning set forth in Section 2.2(c).

"Affiliate" shall have the meaning ascribed to such term in Rule 12b-2 of the Securities Exchange Act of 1934, as amended.

"Agreed Accounting Principles" means the accounting principles set forth on Exhibit D.

"Agreement" shall have the meaning set forth in the preamble of this Agreement.

"<u>Applicable Rate</u>" shall have the meaning set forth in <u>Section 2.2(e)</u>.

"<u>Assets</u>" shall have the meaning set forth in <u>Section 1.1</u>.

"Assumed Contracts" shall have the meaning set forth in Section 1.1(e).

"Assumed Liabilities" shall have the meaning set forth in Section 1.3.

"Benched Employee" shall have the meaning set forth in Section 1.4(i).

"Beneficial Rights" shall have the meaning set forth in Section 1.5.

"Benefit Arrangement" shall mean any plan or arrangement not subject to ERISA maintained or otherwise contributed to by any Seller for the benefit of Business Employees and providing for deferred compensation, bonuses, equity compensation, employee insurance coverage or any similar compensation or welfare benefit arrangement, including without limitation any employment agreement, bonus arrangement, stock option, stock purchase, deferred compensation plan or arrangement, vacation pay, severance pay or other employee fringe benefit plans maintained, sponsored or contributed to by any Seller or any of their Affiliates for the benefit of any Business Employee or dependent or beneficiary of any Business Employee.

"Benefit Plans" shall have the meaning set forth in Section 3.1(m)(i)(A).

"Bill of Sale" shall have the meaning set forth in Section 9.2(c).

"Brown Brothers" shall mean Brown Brothers Harriman & Co.

"Business" shall have the meaning set forth in the recitals to this Agreement.

"Business Consultant" shall have the meaning set forth in Section 3.1(i)(i)(C).

"Business Employee" shall have the meaning set forth in Section 3.1(i)(i)(B).

"Business Permits" shall have the meaning set forth in Section 1.1(f).

"Buyer" shall have the meaning set forth in the preamble of this Agreement.

"Buyer 401(k) Plan Trust" shall have the meaning set forth in Section 5.6(d).

"Buyer Indemnitees" shall have the meaning set forth in Section 8.1.

"Buyer Receivables" shall have the meaning set forth in Section 5.13.

"Buyer's 401(k) Plan" shall have the meaning set forth in Section 5.6(d).

"Buyer's Objection" shall have the meaning set forth in Section 2.2(c).

"Closing" shall have the meaning set forth in Section 9.1.

"Closing Date" shall have the meaning set forth in Section 9.1.

"Closing Statement" shall have the meaning set forth in Section 2.2(b).

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Competing Activities" shall have the meaning set forth in Section 5.7.

"Competitive Business" shall have the meaning set forth in Section 5.7.

"Confidentiality Agreement" means the confidentiality agreement entered into between Troy Gregory and Parent, dated June 22, 2007 (the "Confidentiality Agreement").

"Contracts" shall have the meaning set forth in Section 1.1(e).

"CPA Firm" shall have the meaning set forth in Section 2.2(c).

"Current Assets" shall mean the current assets of the Business that constitute Assets as set forth in the "Working Capital Asset Accounts" on Exhibit D.

"Current Liabilities" shall mean the current liabilities of the Business that constitute Assumed Liabilities as set forth in the "Working Capital Liability Accounts" on Exhibit D.

"DOL" shall have the meaning set forth in Section 3.1(u).

"Encumbrance" shall have the meaning set forth in Section 3.1(f)(i).

"Equity Equivalents" shall have the meaning set forth in Section 3.2(f).

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

"Escrow Agent" shall mean Wells Fargo Bank National Association.

"Escrow Agreement" shall have the meaning set forth in Section 9.2(i).

"Estimated Closing Statement" shall have the meaning set forth in Section 2.2.

"Excluded Assets" shall have the meaning set forth in Section 1.2.

"Excluded Liabilities" shall have the meaning set forth in Section 1.4.

"Excluded Matter" means any one or more of the following: (i) the effect of any change in the United States or foreign economies or securities or financial markets in general; (ii) the effect of any change that generally affects the industry in which the Business operates; (iii) the effect of any change arising in connection with earthquakes, hostilities, acts of war, sabotage or terrorism or military actions or any escalation or material worsening of any such hostilities, acts of war, sabotage or terrorism or military actions existing or underway as of the

date hereof; (iv) the effect of any changes in applicable Laws or accounting rules; or (v) any effect resulting from the public announcement of this Agreement, compliance with terms of this Agreement or the consummation of the transactions contemplated by this Agreement.

"Financing Commitments" shall have the meaning set forth in Section 3.2(e).

"Financial Statements" shall have the meaning set forth in Section 3.1(d).

"GAAP" shall mean generally accepted accounting principles in the United States.

"Government Entities" shall mean any court, arbitrator, department, commission, board, bureau, agency, authority, instrumentality or other body, whether federal, state, municipal, foreign or other.

"<u>Hershey Receivables</u>" shall mean the unbilled and unbilled accounts receivable of Sellers from The Hershey Company or its affiliates prior to the Closing Date.

"Holdings" shall have the meaning set forth in the preamble of this Agreement.

"Indebtedness" means, with respect to any Seller, without duplication, (a) all obligations of such Seller for borrowed money, (b) all obligations of such Seller for the deferred or contingent purchase price of assets, property or services (including without limitation, the current portion of such obligation) and other forms of acquisition-related earnouts, (c) all obligations of such Seller evidenced by bonds, debentures, notes or similar instruments, inclusive of outstanding principal, prepayment premiums, if any, and accrued interest, fees and expenses, related thereto, (d) all obligations under capital leases (which obligations are required to be classified and accounted for as capital lease obligations on a balance sheet of such Seller under GAAP), (e) all obligations, contingent or otherwise, of such Seller as an account party in respect of letters of credit, bankers' acceptances or similar facilities (f) all obligations under any interest rate, currency or similar hedging agreement, (g) all obligations of such person to purchase, redeem, retire, defease or otherwise acquire for value any capital stock or equity interest of such person or any options, rights or warrants to acquire the foregoing, and (h) all direct or indirect guarantee, support or keep well obligations of such Seller with respect to obligations of the kind referred to in clauses (a) through (h) of this definition.

"Indemnification Notice" shall have the meaning set forth in Section 8.3.

"Indemnified Party" shall have the meaning set forth in Section 8.3.

"Indemnifying Party" shall have the meaning set forth in Section 8.3.

"Initial Cash Purchase Price" shall have the meaning set forth in Section 2.1(e).

"<u>Intellectual Property</u>" means any and all worldwide rights in, arising from or associated with the following, whether protected, created or arising under the Laws of the United States or any other jurisdiction or under any international convention: (1) all patents and applications therefor and all reissues, divisions, re-examinations, renewals, extensions,

provisionals, substitutions, continuations and continuations-in-part thereof, and equivalent or similar rights anywhere in the world in inventions and discoveries including, without limitation, invention disclosures ("<u>Patents</u>"); (2) all trade secrets and other proprietary information which derives independent economic value from not being generally known to the public; (3) all copyrights, copyrights registrations and applications therefor ("<u>Copyrights</u>"); (4) all uniform resource locators, e-mail and other internet addresses and domain names and applications and registrations therefor ("<u>URLs</u>"); (5) all trade names, corporate names, logos, slogans, trade dress, trademarks, service marks, and trademark and service mark registrations and applications therefor and all goodwill associated therewith ("<u>Trademarks</u>"); (6) computer programs (whether in source code, object code, or other form), databases, compilations and data, technology supporting the foregoing, and all documentation, including user manuals an training materials relating to the foregoing; and (7) any similar, corresponding or equivalent rights to any of the foregoing anywhere in the world.

"Intercreditor Agreement" shall have the meaning set forth in Section 9.2(g).

"IRS" shall mean the Internal Revenue Service.

"<u>Laws</u>" shall mean any federal, state, local, foreign or other statute, law, ordinance, rule or regulation (including, without limitation, any environmental laws).

"Leased Real Property" shall have the meaning set forth in Section 1.1(a).

"<u>Liabilities</u>" shall mean any direct or indirect liability or obligation that a person owes to or at the behest of any other party, fixed or unfixed, known or unknown, asserted or unasserted, liquidated or unliquidated, secured or unsecured, whether called a liability, obligation indebtedness, guaranty, endorsement, claim or responsibility or otherwise.

"LLC Agreement" shall have the meaning set forth in Section 3.2(g).

"Losses" shall mean (i) all debts, liabilities and obligations owed to or at the behest of any other party; (ii) all losses, damages, judgments, awards, penalties and settlements; (iii) all demands, claims, suits, actions, causes of action, proceedings and assessments, whether or not ultimately determined to be valid; and (iv) all costs and expenses (including interest (excluding prejudgment interest in any litigated or arbitrated matter other than that payable to a third party), court costs and reasonable fees and expenses of attorneys and expert witnesses) of investigating, defending or asserting any of the foregoing, but <u>shall exclude</u>, in each of <u>clauses (i)</u> through (<u>iv</u>) above, consequential, special or punitive damages, including loss of profits, other than such damages awarded to a third party.

"Management" shall have the meaning set forth in the preamble of this Agreement.

"<u>Material Adverse Effect</u>" shall mean an event, occurrence or change that has had or would reasonably be expected to have a material adverse effect on (i) the business, assets, properties, results of operations or financial condition of the Business taken as a whole or (ii) the ability of Sellers to consummate the transactions contemplated by this Agreement, other than in each case an effect resulting from an Excluded Matter.

"Material Contracts" shall have the meaning set forth in Section 3.1(g).

"<u>Net Working Capital</u>" shall mean the amount by which the Current Assets exceed the Current Liabilities, in each case as reflected on the Estimated Closing Statement, the Closing Statement or the Adjusted Closing Statement, as the case may be, and determined in accordance with <u>Section 2.2</u> and the Agreed Accounting Principles.

"Orders" shall mean any order, writ, injunction, judgment, plan or decree of any Government Entity.

"Parent" shall have the meaning set forth in the preamble of this Agreement.

"Parent 401(k) Plan Trust" shall have the meaning set forth in Section 5.6(d).

"Parent Indemnitees" shall have the meaning set forth in Section 8.2.

"Parent's 401(k) Plan" shall have the meaning set forth in Section 3.1(u).

"Permitted Encumbrance" shall have the meaning set forth in Section 3.1(f)(i).

"<u>Pre-Closing Tax Period</u>" means: (i) any Tax period ending on or before the Closing Date; and (ii) with respect to a Tax period that commences before but ends after the Closing Date, the portion of such period up to and including the Closing Date.

"Purchase Price" shall have the meaning set forth in Section 2.1.

"Recent Balance Sheet" shall have the meaning set forth in Section 3.1(d).

"<u>Records</u>" shall mean all lists, records and other information pertaining to accounts, personnel and referral sources, all lists and records pertaining to suppliers and customers, and all books and records of every kind, whether evidenced in writing, electronically (including by computer) or otherwise.

"<u>Registered Intellectual Property</u>" means all United States and foreign: (1) Patents; (2) registered Trademarks, applications to register Trademarks, including intent-to-use applications and other registrations or applications related to Trademarks; (3) Copyright registrations and applications to register Copyrights; (4) URL registrations and applications to register URLs; and (5) any other intellectual property that is the subject of an application, certificate, filing, registration or other document issued by, filed with, or recorded by, any state, government or other public legal authority at any time.

"Sellers" shall have the meaning set forth in the preamble of this Agreement.

"Sellers Intellectual Property" shall have meaning set forth in Section 3.1(j)(i).

"Senior Loan Agreement" means the Loan Agreement between Buyer and Brown Brothers attached hereto as Exhibit D.

"Subordinated Loan Agreement" shall have the meaning set forth in Section 2.1(b).

"Subordinated Note" shall have the meaning set forth in Section 2.1(b).

"Surviving Representations" shall have the meaning set forth in Section 3.3.

"Target Net Working Capital" shall mean an amount equal to \$8,200,000.

"<u>Tax</u>" means any of the following, and "<u>Taxes</u>" means all of the following, imposed by or payable to any Government Entity: any income, gross receipts, license, payroll, employment, excise, severance, stamp, business, occupation, premium, windfall profits, environmental (including Taxes under section 59A of the Code), capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, or value added Tax, any alternative or add-on minimum Tax, any estimated Tax, and any levy, impost, duty, assessment or withholding in the nature of a Tax, in each case including any interest, penalty, or addition thereto, whether disputed or not.

"<u>Tax Returns</u>" means any returns, declarations, reports, claims for refund, information returns or statements relating to Taxes, including any schedules or attachments thereto, and any amendments thereof, to be filed (whether on a mandatory or elective basis) with any Government Entity responsible for the imposition or collection of Taxes.

"Third Party Claim" shall have the meaning set forth in Section 8.4(a).

"Trademark Assignment" shall have the meaning set forth in Section 9.2(f).

"Transfer" shall have the meaning set forth in Section 5.6(d).

"Transfer Date" shall have the meaning set forth in Section 5.6(d).

"Transition Services Agreement" shall have the meaning set forth in Section 9.2(d).

"<u>WARN Act</u>" shall have the meaning set forth in <u>Section 3.1(i)(ii)</u>.

"Warrant" shall have the meaning set forth in Section 2.1(c).

(b) Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

(i) <u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-business day, the period in question shall end on the next succeeding business day.

(ii) Dollars. Any reference in this Agreement to \$ shall mean U.S. dollars.

(iii) <u>Herein</u>. The words such as "<u>herein</u>," "<u>hereinafter</u>," "<u>hereof</u>," "<u>hereof</u>," "<u>hereby</u>" and "<u>hereunder</u>" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

(iv) <u>Including</u>. The word "<u>including</u>" or any variation thereof means (unless the context of its usage otherwise requires) "<u>including</u>, <u>without</u> <u>limitation</u>" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

(v) <u>Number</u>. Where any group or category of items or matters is defined collectively in the plural number, any item or matter within such definition may be referred to using such defined term in the singular number, and vice versa.

IN WITNESS WHEREOF, Parent, Holdings, Management and Buyer have duly executed this Agreement as of the date first written above.

HUDSON HIGHLAND GROUP, INC.

By/s/ Mary Jane RaymondName:Mary Jane RaymondTitle:Executive Vice President and Chief Financial Officer

HUDSON GLOBAL RESOURCES HOLDINGS, INC.

By /s/ Mary Jane Raymond Name: Mary Jane Raymond Title: President

HUDSON GLOBAL RESOURCES MANAGEMENT, INC.

By/s/ Mary Jane RaymondName:Mary Jane RaymondTitle:President

SYSTEM ONE HOLDINGS, LLC

By/s/ Troy GregoryName:Troy GregoryTitle:Chief Executive Officer and President

/s/ Troy Gregory

Troy Gregory, personally solely for purposes of Section 5.11

K-1

For Immediate Release

HUDSON HIGHLAND GROUP

Contact:

David F. Kirby Hudson Highland Group 212-351-7216 <u>david.kirby@hudson.com</u>

Hudson Highland Group Announces Sale of Energy and Engineering Staffing Businesses

NEW YORK, NY – February 4, 2008 – Hudson Highland Group, Inc. (Nasdaq: HHGP) today announced that it has completed the asset sale of its energy and engineering staffing businesses to System One Holdings LLC. The company received as consideration approximately \$11.0 million in cash at closing, subject to post-closing adjustment; a five-year \$5 million note; and a warrant exercisable for 10 percent of the equity of System One. Hudson Highland Group also has the right to receive an additional \$600,000 in cash upon resolution of certain liabilities, and has retained \$3.6 million of receivables of the business. System One is owned and operated by Troy Gregory, former chairman of Hudson Legal in North America.

"We believe this transaction sets up both businesses for greater success," said Jon Chait, Hudson Highland Group chairman and chief executive officer. "For Hudson, consistent with our long-term strategy, it allows us to focus all resources and investment on our high-growth, high-margin professional staffing and solutions businesses. Similarly, System One will have greater control of its own destiny and the ability to develop the infrastructure, tools, employment experience and brand that are best suited to the particular needs of its large-volume engineering resources market."

"The demand for energy today makes this industry one of excellent growth potential. I am looking forward to making System One a great success with its outstanding team of colleagues," said Gregory.

Hudson Highland Group will record a gain on the sale of approximately \$4 million, after recording the note at a discounted value. The company will treat the business as a discontinued operation effective as of December 31, 2007.

For the year ended December 31, 2007, this business reported revenue of \$146.2 million, gross margin of \$18.7 million and operating income of \$3.3 million. For the year ended December 31, 2006, this business reported revenue of \$152.5 million, gross margin of \$20.1 million and operating income of \$5.8 million. For the year ended December 31, 2005, this business reported revenue of \$155.1 million, gross margin of \$22.3 million and operating income of \$7.8 million.

The company expects to offset the impact of this transaction over the next six to nine months through a program to reduce the North American cost structure to match the current size of the business.

Hudson Highland Group

Hudson Highland Group, Inc. is a leading provider of permanent recruitment, contract professionals and talent management services worldwide. From single placements to total outsourced solutions, Hudson helps clients achieve greater organizational performance by assessing, recruiting, developing and engaging the best and brightest people for their businesses. The company employs more than 3,600 professionals serving clients and candidates in more than 20 countries. More information is available at <u>www.hudson.com</u>.

Safe Harbor Statement

This press release contains statements that the company believes to be "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. All statements other than statements of historical fact included in this press release, including those under the caption "Guidance" and other statements regarding the company's future financial condition, results of operations, business operations and business prospects, are forward-looking statements. Words such as "anticipate," "estimate," "expect," "project," "intend," "plan," "predict," "believe" and similar words, expressions and variations of these words and expressions are intended to identify forward-looking statements. All forward-looking statements are subject to risks and uncertainties that could cause actual results to differ materially from those described in the forward-looking statements. These factors include, but are not limited to, the company's history of negative cash flows and operating losses may continue; the ability of clients to terminate their relationship with the company at any time; the impact of global economic fluctuations on temporary contracting operations; risks and financial impact associated with acquisitions and dispositions of non-strategic assets; the company's reliance on information systems and technology; competition; fluctuations in operating results; risks relating to foreign operations, including foreign currency fluctuations; dependence on highly skilled professionals and key management personnel; restrictions imposed by blocking arrangements; exposure to employment-related claims and limits on insurance coverage related thereto; government regulations; and restrictions on the company's operating flexibility due to the terms of its credit facility. Additional information concerning these and other factors is contained in the company's filings with the Securities and Exchange Commission. These forward-looking statements speak only as of the date of this press release. The company assumes no ob

HUDSON HIGHLAND GROUP, INC. UNAUDITED PRO FORMA FINANCIAL INFORMATION

The unaudited pro forma financial information filed herewith as Exhibit 99.2 is incorporated herein by reference and reflects the unaudited pro forma consolidated condensed statements of operations of Hudson Highland Group, Inc. (the "Company") for the nine months ended September 30, 2007 and the year ended December 31, 2006 and the unaudited pro forma condensed consolidated balance sheet of the Company as of September 30, 2007 reflecting the sale (the "Sale") of the Company's energy and engineering staffing business. The unaudited pro forma consolidated condensed statements of operations for the nine months ended September 30, 2007 and the year ended December 31, 2006 give effect to the Sale as if it occurred on January 1, 2006. The unaudited pro forma condensed consolidated balance sheet as of September 30, 2007 gives effect to the Sale as if it occurred on September 30, 2007. The unaudited pro forma financial information is based on the historical financial statements of the Company after giving effect to the Sale and is not necessarily indicative of the financial position or results of operations of the Company that would have actually occurred had the Sale occurred as of January 1, 2006 or September 30, 2007. The unaudited pro forma consolidated condensed financial statements have been prepared based on preliminary estimates. In the opinion of management, all adjustments have been made that are necessary to present fairly the unaudited pro forma financial information. The unaudited pro forma financial information should be read in conjunction with the Company's historical financial statements included in its Form 10-K for the year ended December 31, 2006 and Form 10-Q for the quarter and nine months ended September 30, 2007.

Restatement

As previously disclosed in the Company's Current Report on Form 8-K filed on February 4, 2008, the Company has concluded that it will restate the consolidated financial statements in its previously issued Annual Report on Form 10-K for the year ended December 31, 2006 and Quarterly Reports on Form 10-Q for the quarters ended September 30, 2006, March 31, 2007, June 30, 2007 and September 30, 2007. The Company will include the restated financial information in a filing with the Securities and Exchange Commission prior to or in connection with timely filing its Annual Report on Form 10-K for the year ended December 31, 2007.

The restatement relates to the timing of recording contingent payments related to the acquisition of Balance Ervaring Op Projectbasis B.V. ("Balance") and to expense a portion of the contingent payments, including \notin 1.3 million that was previously recorded as goodwill when the amount was paid in the second quarter of 2007. The restatement results in an increase in the Company's and the Hudson Europe segment's reported operating expenses and an equivalent reduction in EBITDA, operating income, income from continuing operations and net income for each of the periods as follows:

Year ended December 31, 2006	\$1.7 million (\$0.07 per basic and diluted share)
Nine months ended September 30, 2007	\$0.9 million (\$0.04 per basic and diluted share)

The financial information herein for the year ended December 31, 2006, the nine months ended September 30, 2007 and as of September 30, 2007 has been restated to reflect this restatement.

Discontinued Operations

On December 14, 2007, the Company completed the sale of all of the outstanding shares of its Netherlands reintegration subsidiary, Hudson Human Capital Solutions B.V. ("HHCS"). On October 2, 2007, certain of the Company's subsidiaries entered into a purchase agreement to sell Hudson Asia Pacific's trade and industrial business ("T&1"). The financial information herein for the year ended December 31, 2006, the nine months ended September 30, 2007 and as of September 30, 2007 have been restated to reflect HHCS and T&I as discontinued operations. Unaudited pro forma financial information for the sales of HHCS and T&I were included in a Current Report on Form 8-K filed December 19, 2007 and October 29, 2007, respectively.

HUDSON HIGHLAND GROUP, INC. PRO FORMA CONSOLIDATED CONDENSED STATEMENTS OF OPERATIONS (in thousands, except share and per share amounts) (Unaudited)

	For the Nir	For the Nine Months Ended September 30, 2007			
	Restated	Pro Forma Adjustments (A)	Pro Forma Results		
Revenue	\$ 996,372	\$ (107,781)	\$ 888,591		
Direct costs	604,158	(93,496)	510,662		
Gross margin	392,214	(14,285)	377,929		
Operating expenses:					
Selling, general and administrative expenses	361,776	(11,874)	349,902		
Acquisition-related expenses	4,462		4,462		
Depreciation and amortization	11,154	(62)	11,092		
Business reorganization expenses	4,638	—	4,638		
Merger and integration recoveries	(795)	—	(795)		
Operating income	10,979	(2,349)	8,630		
Other income (expense):					
Other, net	3,680	7	3,687		
Interest, net	505	—	505		
Income from continuing operations before provision for income taxes	15,164	(2,342)	12,822		
Provision for income taxes	12,478		12,478		
Income from continuing operations	\$ 2,686	\$ (2,342)	\$ 344		
Earnings per share:					
Basic income per share from continuing operations	\$ 0.11		\$ 0.01		
Diluted income per share from continuing operations	\$ 0.10		\$ 0.01		
Weighted average shares outstanding:					
Basic	25,205,000		25,205,000		
Diluted	25,958,000		25,958,000		

The accompanying note is an integral part of this financial statement.

HUDSON HIGHLAND GROUP, INC. PRO FORMA CONSOLIDATED CONDENSED STATEMENTS OF OPERATIONS (in thousands, except share and per share amounts) (Unaudited)

	For the	For the Year Ended December 31, 2006			
	Restated	Pro Forma Adjustments(A)	Pro Forma Results		
Revenue	\$ 1,310,362	\$ (152,488)	\$ 1,157,874		
Direct costs	831,047	(132,428)	698,619		
Gross margin	479,315	(20,060)	459,255		
Operating expenses:					
Selling, general and administrative expenses	449,839	(14,085)	435,754		
Acquisition-related expenses	1,687	—	1,687		
Depreciation and amortization	19,913	(110)	19,803		
Business reorganization expenses	6,048	(33)	6,015		
Merger and integration expenses	373	(11)	362		
Operating income (loss)	1,455	(5,821)	(4,366)		
Other income (expense):					
Other, net	1,570	14	1,584		
Interest, net	(1,634)		(1,634)		
Income (loss) from continuing operations before provision for income taxes	1,391	(5,807)	(4,416)		
Provision for income taxes	3,771	—	3,771		
Loss from continuing operations	\$ (2,380)	\$ (5,807)	\$ (8,187)		
Basic and diluted per share:					
Income from continuing operations	\$ (0.10)		\$ (0.33)		
Weighted average shares outstanding:					
Basic and Diluted	24,471,000		24,471,000		

The accompanying note is an integral part of this financial statement.

HUDSON HIGHLAND GROUP, INC. PRO FORMA CONSOLIDATED CONDENSED BALANCE SHEETS (in thousands) (unaudited)

	As of September 30, 2007			
	Restated		o Forma stments(B)	Pro Forma Results
ASSETS	Ittotated	<u></u>	<u>otimento(D)</u>	recourte
Current assets:				
Cash and cash equivalents	\$ 34,874	\$	9,063	\$ 43,937
Accounts receivable, net	230,293		(14,229)	216,064
Prepaid and other	16,764		530	17,294
Current assets of discontinued operations	7,340			7,340
Total current assets	289,271		(4,636)	284,635
Intangibles, net	74,685		—	74,685
Property and equipment, net	28,844		(235)	28,609
Other assets	7,407		3,344	10,751
Non-current assets of discontinued operations	1,091			1,091
Total assets	\$401,298	\$	(1,527)	\$399,771
LIABILITIES AND STOCKHOLDERS' EQUITY				
Current liabilities:				
Accounts payable	\$ 30,917	\$	(2,284)	\$ 28,633
Accrued expenses and other current liabilities	129,245		(5,038)	124,207
Short-term borrowings and current portion of long-term debt	15,494		—	15,494
Accrued business reorganization expenses	3,483		_	3,483
Accrued merger and integration expenses	440		—	440
Current liabilities of discontinued operations	4,855			4,855
Total current liabilities	184,434		(7,322)	177,112
Other non-current liabilities	20,205		—	20,205
Accrued business reorganization expenses, non-current	3,763		—	3,763
Accrued merger and integration expenses, non-current	398		—	398
Long-term debt, less current portion	27			27
Total liabilities	208,827		(7,322)	201,505
Total stockholders' equity	192,471		5,795	198,266
	\$401,298	\$	(1,527)	\$399,771

The accompanying note is an integral part of this financial statement.

HUDSON HIGHLAND GROUP, INC. NOTE TO PRO FORMA CONSOLIDATED FINANCIAL STATEMENTS

Note 1. Pro Forma Adjustments

The accompanying unaudited pro forma consolidated condensed financial statements give effect to the following pro forma adjustments necessary to reflect the disposition and discontinuation of operations of the Company's engineering and technical staffing division as outlined in the proceeding introduction as if the disposition occurred on January 1, 2006 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operations and on September 30, 2007 in the unaudited pro forma consolidated condensed statements of operation

- (A) Reduction of revenue and expenses are the result of the disposition of the Company's energy and engineering staffing business. These amounts do not consider an allocation of corporate overhead to the companies that are to be divested, and therefore, selling general and administrative expenses do not reflect any potential reductions in costs in response to this change in the Company.
- (B) Reduction of assets and liabilities are as a result of the disposition of the Company's energy and engineering staffing business and the estimated increase in cash of \$9.1 million from the cash proceeds of \$11.0 million, less other estimated payments and costs related to the disposition of approximately \$1.9 million, the increase in prepaid and other for \$0.6 million deposited into escrow, and a subordinated note in the aggregate principal amount of \$5.0 million discounted to \$3.4 million. The Company retained certain receivables and certain liabilities of the energy and engineering staffing business, which amounts resulted in net assets retained of approximately \$2.1 million at September 30, 2007. Receivables to be retained were approximately \$3.6 million at September 30, 2007.